# The Corporate Responsibility to Respect Human Rights: An Updated Status Review (2022)

Preprin	t · August 2022				
CITATIONS 0	;	READS 2			
2 autho	rs, including:				
15	Claire Methven O'Brien University of Dundee 65 PUBLICATIONS 225 CITATIONS  SEE PROFILE				
Some of the authors of this publication are also working on these related projects:					
Project	Public procurement and human rights View project				
Project	Business and Human Rights View project				

# The Corporate Responsibility to Respect Human Rights:

# **An Updated Status Review (2022)**

Claire Methyen O'Brien and Giulia Botta

To be published in João Luiz da Silva Almeida (ed), *Responsabilidade Social Corporativa e Governança Socioambiental: Greenwashing e Direitos Humanos* [Corporate Social Responsibility and Social and Environmental Governance: Greenwashing and Human Rights] (Rio de Janeiro, Lumen Iuris, manuscript submitted, forthcoming 2022).

Abstract: The United Nations (UN) Human Rights Council's Protect, Respect and Remedy ('Three Pillar') Framework on Business and Human Rights and the Guiding Principles on Business and Human Rights (GPs) represent a significant marker in the contemporary evolution of norms on the responsibility and accountability of corporate actors for their social, environmental and human rights impacts. Building on earlier work, this chapter provides an updated exposition and critical assessment of developments since 2011 in law, policy and practice that have sought, in line with the GPs, to advance corporate respect human rights. First the chapter briefly addresses the origins of the GPs and their standing in relation to wider principles and standards of international human rights law (Section 1). It then explains the notion of corporate human rights due diligence as the principal mechanism through which businesses are envisaged, by the GPs, to realise respect for human rights. Here it relates measures being taken by governments and others to legalise corporate human rights due diligence requirements, such as due diligence and supply chain laws recently adopted in France, Germany, Norway and prospectively by the European Union (Section 2). In Section 3, the chapter addresses in turn three elements of the corporate human rights due diligence process: human rights impact assessment (HRIA, Section 3.1); companies' integration of the findings of human rights risk identification and the exercise of leverage (Section 3.2); and corporate human rights reporting (Section 3.3), where we address links to 'ESG' investment practices. In conclusion, given the continuing scale and severity of business-related human rights abuses, but also uncertainties about design and impacts of reviewed regulatory measures, the chapter calls for intensified scholarly inquiry and debate in corporate responsibility research.

**Keywords:** Human rights, responsible business conduct, supply chain, value chain, due diligence corporate responsibility to respect human rights, UN Guiding Principles on Business and Human Rights, European Union, Council of Europe, non-financial reporting, ESG, trafficking, conflict minerals

**JEL Classification:** K29, K33, K30, K10, K20, F60, L86

Claire Methven O'Brien is a Baxter Fellow and Lecturer, University of Dundee; Senior Researcher, Danish Institute for Human Rights; Global Practitioner, University of Strathclyde Business School. Email: <a href="mailto:cobrien001@dundee.ac.uk">cobrien001@dundee.ac.uk</a>.

**Giulia Botta** is a Doctoral Candidate in Law & Sustainable Development, University of Milan; Visiting Fellow at the Danish Institute for Human Rights and at the Centre for Private Governance, Copenhagen University. Email: <a href="mailto:giulia.botta@unimi.it">giulia.botta@unimi.it</a>.

# The Corporate Responsibility to Respect Human Rights:

An Updated Status Review (2022)<sup>1</sup>

Claire Methven O'Brien and Giulia Botta

### 1. Introduction

The United Nations (UN) Human Rights Council's Protect, Respect and Remedy ('Three Pillar') Framework on Business and Human Rights and the Guiding Principles on Business and Human Rights (GPs) represent a significant marker in the contemporary evolution of norms on the responsibility and accountability of corporate actors for their social, environmental and human rights impacts (UN Human Rights Council (UNHRC) 2008, UNHRC 2011)

From an ethical viewpoint, international human rights standards such as the Universal Declaration of Human Rights (UN, 1948) and subsequent international and regional human rights instruments ought to have provided a natural frame for increasingly frequent attempts, during the 1980s and 1990s, to seek redress for business-related infractions of human dignity, fundamental interests, well-being and welfare (Dhanarajan and Methven O'Brien, 2015). Yet legal as well as political obstacles obstructed the use of human rights treaties and courts to challenge corporate misconduct.

Top amongst these, historically, was the limited application of human rights obligations to non-state actors and in the "private sphere" (Clapham, 1996). In general, international human rights law does not establish directly enforceable duties on businesses, even if corporations have indirect responsibilities to comply with substantive standards of protection entailed by human rights through requirements imposed on them via national laws, for instance, in the areas of employment and environment (Methven O'Brien, 2019).

However, such national laws, as well as contractual and investment relationships, may not be fully aligned with international human rights standards, while companies do not always comply with them in any case. During the 1990s concerns grew, furthermore, that liberalisation, market integration and 'regulatory competition' on the global plane were exerting downwards pressure on national social and environmental standards (Davies and Vadlamannati, 2013). Mounting frustration at lack of effective accountability for business-related human rights abuses for victims, in face of the expanding rights and power of corporations, correspondingly triggered popular protests, such as the 1999 'Battle of Seattle', and eventually international political concern (Dhanarajan and Methven O'Brien, 2015; Methven O'Brien, 2022). Many businesses, meanwhile, deferred responding to such disquiet: even by 2010, still 42.8% of FTSE100 policies did not address human rights at all (Preuss and Brown, 2012).

In 2005, UN Secretary General Kofi Annan appointed John Ruggie as his Special Representative on Business and Human Rights (SRSG) with a mandate to identify and clarify standards of corporate

\_

<sup>&</sup>lt;sup>1</sup> Parts of this paper draws on an earlier article: Claire Methven O'Brien and Sumithra Dhanarajan, "The Corporate Responsibility to Respect Human Rights: A status review", *Accounting, Auditing and Accountability Journal Vol. 29, Issue 4, April 2016, pp. 542-567.* 

responsibility and accountability for human rights. An earlier UN-level attempt to transact human rights standards for business had failed to garner cross-stakeholder support. The UN Sub-Commission on the Promotion and Protection of Human Rights had, in its 2003 UN Draft Norms on the Responsibilities of Transnational Corporations and Other Business Enterprises with regard to Human Rights (UN Human Rights Commission 2003) sought to specify legal human rights duties for companies that were broadly co-extensive with those of governments (Methven O'Brien 2009).

Ruggie rejected this approach from the outset. Rather, in pursuit of what he often labelled "principled pragmatism" he emphasised human rights as imposing ethical and social expectations of business (Ruggie 2013). Such expectations, he maintained, were implicit in existing human treaties, so that new international laws to establish corporate duties to uphold human rights were not needed. Accordingly, the 2008 UN Framework and 2011 GPs advanced by Ruggie maintained the state's role as primary bearer of the duty to protect human rights as their first 'pillar'. Alongside, however, they asserted a corporate responsibility to respect, and not harm, human rights, as a second pillar. To give effect to this responsibility, companies should deploy dedicated corporate policies and internal measures, encapsulated in the notion of process, termed "human rights due diligence". Finally, the UN Framework's third "pillar" recognised the right to effective remedy of victims of business-related human rights abuses and corresponding need for accessible remedy mechanisms (UNCHR, 2011)).

Ruggie assessed that this approach, amplifying but remaining anchored in established international legal principles, could more readily find support by states and businesses than new standards tangential to them. Assimilating the UN framework and GPs into government and corporate policy and practice, he claimed, would "reduce corporate-related human rights harms to the maximum extent possible in the shortest possible period of time" (Ruggie, 2008). Arguably supporting this view, international organisations (OECD, 2011 and 2018; COE, 2018; IACHR, 2019, European Commission, 2011), large businesses, and governments, since 2011 have espoused the GPs and undertaken a multitude of initiatives to advance their institutionalisation including, for instance, new laws on due diligence and corporate human rights reporting, multi-stakeholder processes to develop National Action Plans on business and human rights (Methven O'Brien *et al* 2014, Methven O'Brien *et al* 2015; Methven O'Brien Ferguson and McVey 2022) and a multitude of awareness-raising, educational and capacity-building initiatives within industry sectors, individual firms and civil society organisations.

Accordingly, the GPs may have helped to shore up the legitimacy of both market integration and human rights, in the 2010s, by re-orienting human rights norms, if not international laws, to address contemporary threats to individuals, societies and the environment deriving from business activities. Yet doubts have remained about the sufficiency of the GPs' approach (Methven O'Brien, 2011; Deva 2012; Černič and Van Ho, 2015). Data consistently question the GPs' penetration and practical effectiveness. To illustrate, even when a majority of corporate leaders conceded holding responsibility for human rights, only a fifth of their companies had a human rights policy, according to a 2015 study (EIU 2015). In a 2017 analysis, almost half of company respondents were found never to have undertaken human rights due diligence or impact assessment (BIICL, Norton Rose Fulbright LLP, 2017). By 2020, still almost half of the world's largest companies could not demonstrate measures to address human rights risks in their supply chains (Corporate Human Rights Benchmark 2020).

Besides, business practices self-evidently harmful to human rights persist (ILO2020, OHCHR 2022a, OHCHR 2022b) while new ones have continued to emerge, for instance, in the tech sector (Methyen

O'Brien, Jørgensen, and Hogan 2021). If obstacles to remediation for victims of corporate human rights harms are now better understood, neither have they yet been dismantled (OHCHR, 2016; EU FRA, 2020). While the GPs' rhetoric may have captured the policy-making "peaks", then, their slow uptake on the ground seems to substantiate enduring truth in the claim that, in many cases, "firms are still not ready to be safe rather than sorry" (Aaronson and Higham, 2013, p. 333).

For some authors, this shows the need for more prescriptive international laws, including a binding UN human rights treaty to address duties directly to companies (Bilchitz, 2013). But would this necessarily yield better results than the GPs, in terms of increased implementation, enforcement and effectiveness of human rights in the business setting? Scholarship indicates many disparate internal and external factors as promoting CSR (Lim and Tsutui 2012; Lim 2017; Pope and Lim 2020) or frustrating compliance, beyond the bare legal or non-legal character of the rules in question (Gunningham *et al.*, 2003) while continuing corporate scandals in highly regulated areas seem to bear this out (Russell *et al*, 2016; Methven O'Brien 2022). Governments' respect for human rights is hardly perfect, even where they are obligated under by international treaties or national laws, as many scholars have observed (Hathaway 2002; Neumayer 2005, Goldsmith and Posner 2005, Hathaway 2007, Moyn 2012, Posner 2014; Ford and Methven O'Brien 2017).

If the matter is not settled in social science, neither is its debate over politically. While many states continue to see virtue in the GPs' 'soft law' approach (UNHRC 2014a), others have launched a process to elaborate a new business and human rights treaty (UNHRC, 2014b). For the moment, this global initiative to 'legalise' corporate responsibility for human rights appears to have stalled (Methven O'Brien 2021, Subasinghe, 2021, Botta, 2022). Yet at national and regional levels, some steps are being taken to formalise business responsibilities for human rights in law, and more are in pipeline. Such efforts embrace the enactment of statutory duties on companies to perform human rights due diligence as well as indirect approaches, such as human rights-related disclosure and investor requirements (Methven O'Brien 2020a, 2020b). While potentially far-reaching, the ultimate impacts of this new regulatory mosaic on companies, their conduct, and consequences for rightsholders, governments and markets for now remain unknown.

Building on earlier work (Methven O'Brien and Dhanarajan, 2015), this chapter reviews selected developments since 2011 to consider how the corporate responsibility to respect human rights expressed in the GPs has been interpreted and implemented through law, policy and business behaviour. Section 2 outlines the iterative process of due diligence identified by the GPs as a principal mechanism by which business respect for human rights is intended to be rendered operational. It next highlights new legal due diligence requirements and, relatedly, persisting ambiguities around the character, functions and legal consequences of corporate human rights due diligence procedures. Section 3 deals in more depth with three of due diligence's subsidiary components: human rights impact assessment (section 3.1); integration of the findings of human rights risk identification and impact assessment into company policies and practices (section 3.2); and corporate human rights reporting (section 3.3). Each part identifies expectations of companies under the GPs while also highlighting challenges and dilemmas with reference to cross-disciplinary scholarship, primary legal and policy materials. Section 4 discusses the effects and effectiveness of legal due diligence requirements, with reference to available evidence, while highlighting gaps in knowledge attendant on the still emergent character of this new regulatory field. Section 5 concludes.

# 2. Human rights due diligence in the GPs and related legal requirements

According to the GPs, a 'corporate responsibility to respect human rights', implied by human rights treaties, requires businesses both to avoid infringing human rights and to address adverse human rights impacts in which they are involved. Businesses should seek to prevent or mitigate impacts that they have "caused or contributed to", as well as those "directly linked" to their operations, products or services through their business relationships, whether contractual or non-contractual (UNHRC 2011, GP13). In terms of subject-matter, the corporate responsibility to respect human rights refers to all internationally-recognised human rights, not just those that are at a given moment binding in any one jurisdiction (UNHRC 2011, GP11).

"Human rights due diligence" as described by the GPs is a cyclical process through which, if they execute it correctly, corporations can fulfil their responsibility to respect human rights. Its first step is adopting a policy to respect human rights (GP15). A high-level statement expressing company commitment to respect human rights is pre-requisite to effective due diligence: board-level 'buy-in', the GPs suggest, is critical to on-the-ground follow-through, especially in the face of conflicting business imperatives. Company human rights policies should furthermore be public, to give external stakeholders a clear platform for engagement with, and scrutiny of, companies whose activities affect them (UNHRC 2011, GP16).

After adopting a human rights policy, due diligence is envisaged as comprising four further steps, in form resembling an archetypal continuous improvement cycle (UNHRC 2011, GPs17-20):

- 1) Assessing actual and potential impacts of business activities on human rights ("human rights risk and impact assessment");
- 2) Acting on the findings of this assessment, including by integrating appropriate measures to address impacts into company policies and practices;
- 3) Tracking how effective the measures the company has taken are in preventing or mitigating adverse human rights impacts; and
- 4) Communicating publicly about the company's due diligence process and its results. In light of the right of victims of business-related human rights abuses to access effective remedies, constituting the "third pillar" of the UN Framework, companies should also take steps to remediate adverse impacts of their activities on rights-holders (UNHRC 2011, GP22).

In terms of scope, aligning with the 'corporate responsibility to respect' from which it derives, due diligence should be addressed to business impacts on all human rights enumerated in the International Bill of Human Rights (UN General Assembly 1948, UN General Assembly 1966a; UN General Assembly 1966b) and the labour standards contained in the International Labour Organisation's Declaration on Fundamental Principles and Rights at Work (ILO, 1998) at a minimum. Based on businesses' specific circumstances, additional standards, such as those relating to indigenous peoples (ILO, 1989; UN General Assembly, 2007) or conflict-affected areas (UNHRC 2011, GP12) may also be relevant. This breadth does not however imply that due diligence becomes unwieldy. According to the GPs, it remains practicable because companies are permitted to adjust the scale and intensity of the due diligence exercise to their individual character and context, taking into account factors such as company size, industry sector, and the seriousness and extent of human rights impacts to which the company's activities may give rise (UNHRC 2011, GP14).

Since 2011, numerous governance actors have endorsed the GPs explicitly and called on companies to adopt them and apply due diligence. In 2011, for example, the European Union did so in its

Communication on CSR (European Commission, 2011) while in 2014 the UN Human Rights Council followed suit (UNHRC, 2014; Methven O'Brien, Mehra, Blackwell, Poulsen-Hansen 2016). National Action Plans on business and human rights, of which 42 have been adopted globally, with several more in pipeline (DIHR n.d.) likewise generally call on companies to respect human rights and do due diligence as provided for in the GPs (OHCHR, 2016; Methven O' Brien, Ferguson, McVey, 2021).

Besides, taking point of departure in the GPs, governments and others have promoted various norms of their own that encourage or require companies to implement a GPs-style human rights due diligence process, elements of it, or closely related practices. At first, such efforts comprised guidance, for example, the OECD's 2011 revision of its Guidelines for Multinational Enterprises (MNEs) (OECD, 2011) and its subsequent Due Diligence Guidance for Responsible Business Conduct (OECD, 2018). The Equator Principles (EP, 2003), similarly, were revised with an aim to align with the GPs (EP, 2020). While technically non-binding, such standards are still capable of giving rise to practical consequences as well as legal duties, if incorporated voluntarily into corporate policies, purchase contracts (Cutler 2017; Martin Ortega 2018; Göthberg 2019), investment or lending agreements (DIHR 2019, OHCHR, 2021), for example. In some jurisdictions, including Colombia and Peru, such 'voluntary' standards may also be given limited judicial application, for instance, as an aid to interpretation of constitutional provisions (Sanabria and Schonfelder 2021, Corte Constitucional 2016; Debevoise & Plimpton, 2021; SUTREL, n.d.)., with some suggestion of scope for such deployment also in the Inter-American human rights system (*Miskito Divers (Lemoth Morris et al) v Honduras* IACHR, 2021).

Despite such possibilities for legal traction, evidence has continued to suggest that adherence to the GPs and due diligence amongst businesses remains marginal, even in high-risk sectors (Business and Human Rights Resource Centre, 2022). A 2013 study that assessed 200 randomly-selected large European companies found, for instance, that only 33% referred to GPs-aligned standards (UN Global Compact, OECD Guidelines for Multinational Enterprises, or International Standards Organisation's (ISO) ISO 26000 Social Responsibility guidance (ISO 2010)); only 3% to the UNGPs themselves; and 2% to the ILO's Tripartite Declaration of Principles concerning Multinational Enterprises and Social Policy (MNE Declaration) (ILO 2006; European Commission, 2013). A survey of 186 European companies in the same period found that over 80 per cent did not provide public information on checks of their supply chains for linkages to conflict or human rights abuses (Steinweg and ten Kate 2013). By 2016, there were still only three-hundred and fifty corporate human rights policies amongst approximately 80,000 transnational firms worldwide (Business and Human Rights Resource Centre 2016). Perhaps more optimistically, in a 2021 survey of 760 Japanese companies listed on the Tokyo Stock Exchange, 69 per cent said they had formulated a specific policy on respecting human rights or had clearly stipulated it in their company policies, management principles and strategies; 52 per cent replied that they had conducted human rights due diligence and initiated investigations to safeguard human rights. Still, even here, more than one-fifth (21 per cent) expressly replied that they had not taken steps towards GPs implementation (METI, 2021; Human Rights Resource Center, 2021).

At the same time, some early studies indicated potential for the GPs to influence business perceptions and risk analysis where companies did engage with the GPs internally (MacPhail and Adams, 2016, McCorquodale, Smit, Neely, Brooks, 2017). Alongside persisting abuses, this put pressure on governments to 'upgrade' due diligence from a social expectation to a legal requirement. While advocates have sought in this context to extend prevention and accelerate remediation of abuses,

amongst companies, "early adopters" of human rights policies have welcomed the prospect of due diligence legislation as a mechanism to 'level the playing field' with less-compliant competitors (DIHR, ICAR and GBI, 2014; BusinessEurope 2022; Business and Human Rights Resource Centre 2019).

Legal due diligence requirements started to appear at first in indirect form, via duties on companies to report on due diligence processes as featured, for example, within non-financial reporting (EU 2014) and modern slavery statutes (State of California, Department of Justice, 2012; United Kingdom of Great Britain and Northern Ireland, 2015, Australian Government, 2018). Admittedly, some such requirements took the form of 'comply or explain' duties that can barely be reckoned as 'hardening' corporate human rights due diligence into a legal obligation (EU, 2014; United Kingdom of Great Britain and Northern Ireland, 2015). In Brazil, a differently styled, though still disclosure-based measure, intended to encourage company engagement with human rights issues on a thematic basis, is the so-called 'Dirty List' of employers publicly named by authorities to have subjected workers to conditions amounting to modern slavery (ESCR-Net CAWG, 2020).

Subsequent initiatives have however sought to legalise the UNGPs' due diligence concept more directly, firstly, via thematic or sector-specific laws, or alternatively, through 'horizontal' due diligence requirements. Examples of the former concern conflict minerals. Due diligence provisions already featured in s1502 of the US Dodd-Frank Act (US Congress 2010; US Securities and Exchange Commission, 2012), requiring companies to undertake due diligence to verify the presence of such minerals within their products, as well as publicly to disclose their due diligence reports. The EU Conflict Mineral Regulation (European Commission 2017) established due diligence, traceability and disclosure requirements for approximately 600-1,000 EU importers of tin, tantalum, tungsten and gold from conflict-affected and high-risk areas, with reference to the OECD's Due Diligence Guidance for Responsible Supply Chain of Minerals concerning the sourcing of natural resources from conflict-affected and high-risk areas (OECD, 2013; OECD 2016). More specifically, under the Regulation, companies must have a 'chain of custody or supply chain traceability system' that evidences, for imported minerals and metals, their country of origin, date of extraction, supplier, and additional data specified for the substance in question in the OECD Guidance (European Commission 2017, art. 4). The EU Regulation was preceded by national laws passed by the Congolese and Rwandan governments, in the early 2010s, requiring companies in the tin, tantalum, tungsten or gold mining sectors to undertake supply chain due diligence also according to the OECD standard, making these jurisdictions amongst the first to enact national due diligence laws (Global Witness, 2014). As a related development, in 2015, the Chinese government launched draft Chinese Due Diligence Guidelines for Responsible Mineral Supply Chains setting out a five-step process whereby companies review their supply chains for specific risks and publish detailed information on actions taken (China Chamber of Commerce of Metals, Minerals and Chemicals Importers and Exporters, 2015).

Other examples of commodity, product or sector-specific laws include the EU Timber Regulation (EU 2010; European Commission 2021), requiring due diligence by EU traders who place timber products on the EU market; a similar Australian law (Australian Illegal Logging Prohibition Act 2012); and a proposed EU Batteries Regulation that would introduce mandatory requirements for all batteries placed on the EU market to ensure their production does not lead to human rights abuses or environmental damage (EU Commission, 2020). Thematic due diligence laws and other measures are also emerging, for example, on child labour and conflict minerals in Switzerland (Schweizerische Eidgnenossenschaft, 2021, Bueno and Kaufmann, 2021), and on forced labour in the US. The Uyghur Forced Labor Prevention Act (US Congress, 2022, S.65) foresees a Strategy to Prevent the

Importation of Goods Mined, Produced, or Manufactured with Forced Labor in the People's Republic of China (US Congress, 2022) that includes due diligence obligations related to assessing, preventing and mitigating forced labour risk in the production of goods imported into the United States as well as associated guidance (Littenberg, Elliot, Witschi, 2022).

'Horizontal' due diligence requirements appear more similar in form to the expression of due diligence in the GPs, lacking sector or product-based restrictions in their scope of application. Such duties feature, for instance, in France's 2014 Loi de Vigilance (République Française, 2017) and laws passed in Germany (Bundestag 2021) and Norway (Stortinget, 2021) in 2021. A draft corporate human rights and environmental due diligence directive in somewhat like terms is under consideration in the EU (European Commission, 2022, Methven O' Brien, Martin-Ortega, 2022). Under this proposal, large limited liability companies, including EU and third-country companies operating in the EU market and smaller companies in designated 'high-risk' sectors (European Commission, 2022, art. 2) would be required to identify, prevent, mitigate and account for their adverse human rights impacts, and also their environmental impacts, in their own operations but also their subsidiaries, contractors and subcontractors, to the extent of their "established business relationships". Under the draft EU Directive, harms resulting from failures of due diligence might attract administrative sanctions or exposure to civil liability in proceedings brought by damaged parties (Methven O'Brien, Martin-Ortega 2022). According to a 2020 survey, a majority (73.37 per cent) of large businesses stand to support such a horizontal EU approach, viewing the existing patchwork of indirect and thematic laws as lacking effectiveness, efficiency and coherence (Smit et al, 2020). Proposals for mandatory corporate human rights due diligence legislation are also emerging in the Americas. In Mexico, a draft General Law on Corporate Responsibility and Due Diligence was presented in 2020 following a recommendation by the country's National Human Rights Council the previous year (OHCHR, 2018). In Brazil, draft legislation has been published to succeed earlier formal guidelines (Serva, Faria, 2022; Thame, 2019; Presidência da República, 2018 - Decree No 9.571/2018; National Human Rights Council of Brazil, 2020 – Resolution No 5/2020)).

Notably, however, horizontal due diligence laws seen to date diverge from the formulation of the responsibility to respect human rights and due diligence under the GPs in one way or another, whether in terms of limiting the personal scope, depth or breadth of due diligence requirements. They also embody significant discrepancies *inter se*, for instance, regarding their personal and subject matter scope, the question of how far down the supply or value chain their requirements extend and regarding what roles 'safe harbour' provisions and civil liability should play within their legislative schemes (Methven O'Brien and Martin-Ortega 2022).

This may not be surprising, given that the UNGPs' loose definition of due diligence was not intended to serve as a template for legislation. Moreover, given John Ruggie's conscious election in the GPs to express a broad corporate responsibility to respect human rights and due diligence as a social expectation, that only crystallised in the dynamic context and relations of the individual corporation, rather than in a legal duty (Ruggie and Sherman 2017; Bonnitcha and McCorquodale 2017a; Bonnitcha and McCorquodale 2017b), the enactment of due diligence laws evidently entails an interesting qualitative transformation (Backer 2017). The character, implications, and associated dilemmas of this transformation into law, are addressed further in Section 4 below. As a prerequisite to that discussion, the following section explores further three key aspects of the due diligence process.

# 3. Due diligence unpacked: impact assessment, integrating findings and reporting

# 3.1 Human rights impact assessment

Human rights impact assessment (HRIA) is the first step in the GPs' due diligence cycle. An adverse human rights impact may be said to occur when an action removes or reduces the ability of an individual to enjoy his or her human rights (International Business Leaders' Forum and International Finance Corporation, 2010). An HRIA, then, provides an overview of actual or potential human rights impacts posed by a company's business activities. Companies can be connected to such impacts in a variety of ways, directly or indirectly. According to the GPs, companies may be responsible for: a) causing a human rights impact through intended or unintended actions (for example, deliberate discrimination in hiring practices, or accidental pollution of a local waterway); b) contributing to a human rights impact, by being one of a number of entities whose conduct together curtails human rights (for instance, where a lead purchaser changes its order specifications at short notice so that its suppliers breach labour standards in fulfilling it); or c) impacts directly linked to a business' operations, products or services, through its business relationships, including those with suppliers, joint-venture partners, direct customers, franchisees and licensees (UNHRC 2011, GP13).

The GPs indicate that companies should, in the course of performing an HRIA, draw on internal or independent human rights expertise; undertake meaningful consultation with potentially affected rights-holders and other relevant stakeholders; consider human rights impacts on individuals from groups that may be at heightened risk of vulnerability or marginalisation, and gender issues; and repeat risk and impact identification at regular intervals, for instance, before entering into a new activity, prior to significant decisions about changes in activities, and periodically throughout a project life-cycle (UNHRC 2011, GP18).

Although the GPs' guidance on HRIA remains high-level, a multiplicity of guidance, tools and methods have been advanced by other actors further articulating the steps of screening, planning and scoping, data collection and baseline development, analysing and disaggregating impacts, impact mitigation and management, monitoring, reporting and evaluation (Abrahams, Wyss, 2010; World Bank, 2013; Gotzmann, 2019; UNHRC 2019; DIHR, 2020). Securing effective participation of rights-holders in the HRIA process and follow-up is often given particular emphasis in this context (Columbia Center on Sustainable Investment et al, 2017; Buhmann 2020).

Many sector-specific HRIA resources have also been developed (International Council on Mining & Metals (ICMM), 2012; IPIECA and DIHR, 2013; NomoGaia, n.d., DIHR 2020) as well as thematic HRIA guidance focusing on the rights of children (UNICEF 2014, Save the Children, 2016) and indigenous peoples (IBIS Denmark, 2013, DIHR 2019), for instance. In some cases, companies have devised HRIA methodologies tailored to their own operating environments (Wyss and Bensal 2019). To date, however, few HRIAs have been published (Goldcorp, 2010, Nestlé and the Danish Institute for Human Rights, 2013) making their prevalence hard to ascertain; this may change as new due diligence reporting requirements take effect, see further section 3.3 below). Company HRIAs that have been disclosed have, on the other hand, frequently met with criticism *inter alia* for being too restrictive in scope or in terms of the adequacy of the process adopted (Salcito, 2019; DeWinter-Schmitt and Salcito 2019). Thus, civil society organisations and national human rights institutions (NHRIs) have also undertaken HRIAs (Hamm et al., 2013) that go beyond corporate practice, for instance, in terms of involving rights-holders, an as regards transparency and disclosure (International Federation for Human Rights, 2014; DIHR, 2016; International Centre for Human Rights and Democratic Development, n.d; Tamir and Zoen 2017), with the aims to highlight shortcomings of

specific business HRIAs and more generally to demonstrate the availability of alternative methodologies.

The parameters of HRIA under the GPs hence remain somewhat emergent and contested (Gotzmann 2019). One question is whether HRIA should be integrated into environmental or social impact assessment processes, or undertaken as a separate, "stand-alone" exercise (IPIECA and DIHR 2013, DIHR 2016). As climate change concerns have grown (OHCHR nd), it has been urged that this issue too should be included in the scope of due diligence and hence also impact assessment (Bright and Buhmann 2021) potentially adding further to the analytical complexity of the HRIA exercise. How to achieve neutrality in the conduct of HRIA, given power asymmetries as well as often conflicting interests between companies, communities and public authorities is another enduring challenge (Schilling-Vacaflor, 2012) that has prompted support for 'hybrid' company-community approaches (Tamir and Zoen 2017). Additional questions relate to the potential of "strategic" or sector-wide HRIA, mirroring environmental practice (Wachenfeld, Wrzoncki, and de Angulo 2019); the role in HRIA of human rights indicators (Engle Merry, 2011; Veiberg, Factor and Tedaldi, 2019; Maher 2020); while some have queried the sense of risk-based approaches (Power, 2004; Taylor et al., 2009), and the notion of "impact" assessment itself (Boele and Crispin, 2012). Such matters are likely to grow further in prominence with the inclusion of references to HRIA, and its linkage to liability for corporate harms, in new due diligence laws (Schilling-Vacaflor 2020; Methven O'Brien and Martin-Ortega 2022).

# 3.2. Businesses responses to human rights risks and leverage

Once they have identified and assessed their human rights impacts, the GPs call for businesses to respond by preventing future abuses and addressing any that have been uncovered. Clearly, the potential range of actions required of companies in this regard is broad. Internally, a company might need to amend recruitment processes or contractual terms for employees; change its purchasing, subcontracting, sales or marketing practices; improve worker accommodation; introduce due diligence for land acquisitions; or, in the tech sector, address content moderation, or AI development or applications, for example. In addition, implementing such changes will usually require the allocation of new resources by the business in question and potentially other actors such as suppliers, for instance, for training and awareness-raising, monitoring and management of human rights impacts on a continuous basis (Gap Inc., n.d.; Nestlé Global, n.d.). Businesses are expected to address all their actual or potential impacts, though the GPs foresee they may need to prioritise amongst them: it is recommended that companies first seek to prevent and mitigate their severest impacts, or those where a delay in response would make consequences irremediable (UNHRC 2011, GP24).

Under the GPs, where impacts are *caused* by elements within the business itself, it should cease or prevent the impact, and provide for, or collaborate in, remediation. If a company has *contributed to* or is *directly linked* to impacts, it should cease its own contribution, exercise 'leverage' over other entities involved, and provide, or cooperate in, remediation. According to the GPs, 'leverage' is a company's ability to effect change in the wrongful practices of another entity, be that a business, a public or other social actor, with which it has a relationship. Modalities of leverage are thus numerous, ranging from capacity building, formal representations or informal lobbying, to amending contract terms for suppliers (Shift, 2013a). If a company has leverage over a business partner, it is expected to exercise it. If, on the other hand, the company lacks leverage, it is expected to seek ways to increase it, for example, by offering incentives, or applying sanctions to the relevant entity, or collaborating with others to influence its behaviour. Where risks or impacts derive from a company's business relationships, rather than from its own activities, the GPs further require it to consider, in determining

appropriate remedial measures, *inter alia* how crucial the relationship is; the severity of the actual or possible abuse; and whether terminating the business relationship would itself have adverse human rights consequences (UNHRC 2011, GP19; cf. Wood, 2011).

If the GPs' notion of leverage appears straightforward, views nevertheless vary on its practical application. Banks, for example, have tended to emphasise constraints on their leverage over those they lend to (Thun Group of Banks, 2013). Outsiders on the other hand argue that they wield substantial influence over clients (BankTrack and The Berne Declaration, 2013; Myerstein, 2013; Backer 2017) and can 'piggy-back' human rights onto other due diligence obligations to which they are already subject in many jurisdictions, for instance, concerning anti-corruption (Finance Against Trafficking, 2014; UNEP Finance Initiative, 2014; Van Dijk, De Haas and Zandvliet, 2018).

Likewise, investors may be regarded as well placed to use leverage to force companies to implement human rights due diligence as a preventative tool (UNHRC, 2021; Aizawa et al, 2018, McCorquodale, Nolan, 2021). But even relatively simple issues such as whether minority shareholders are "directly linked" to investee's adverse human rights impacts for the purposes of the GPs, and therefore expected to exercise leverage, initially attracted disagreement. Two complaints ("specific instances") on this point were lodged under the OECD Guidelines for MNEs (OECD, 2011) in relation to investments by Norway's Government Pension Fund Global and the Dutch Pension Fund in a proposed iron mining development, yielding the conclusion that a minority shareholding does entail investor responsibility to identify, and hence prevent, negative impacts, this finding being later upheld in guidance (OECD, 2014, Ruggie and Nelson, 2015).

Generally, though, the GPs' application to complex and dynamic financial products and relationships seems likely to pose continuing uncertainties and dilemmas, while the impact of investor 'engagement' on human rights also remains unclear (Kolstad 2016). In 2020, investors representing over USD 4.5 trillion in assets under management wrote to the worst-performing 95 companies on the CHRB calling for urgent improvements. Of those companies, only 16 improved their ranking in the next year (McCorquodale 2020) arguably providing to support to those who emphasise divestment over investor 'engagement', and more general critiques of governance gaps presented by the 'financialisation' of sustainability goals (Hiss 2013, Park 2018a). Similar doubts appear warranted regarding the actual human rights impacts of social 'impact' investments (Park 2018b).

Another issue concerns companies' ability to influence the use of their products or services by customers (OECD Watch, 2014). This is relevant, for instance, with regard to policing and military supplies, surveillance equipment and other dual use technologies (Cohn et al., 2012; Wagner, 2012; Frediani, 2014; Wagner and Guarnieri, 2014). Though the export of such products may be permissible under national standards, the GPs require companies to look beyond formal legality to ascertain whether their products or services facilitate human rights abuses (Marotta, 2013). This will be easier to ascertain in some situations more than others (Amnesty International 2019; Schliemann and Bryk 2019). Likewise preventing the use of social media platforms as a medium for hate speech and the organisation of criminal acts is a pressing issue (Council of Europe, 2014; Essers, 2014; Jørgensen, Methven O'Brien and Hogan 2020). Defining the responsibility of internet service providers and achieving transparency, control and accountability over how they balance freedom of expression and other rights and prerogatives is a question not yet fully answered in principle (Zuboff 2019), law, or practice (Klonick 2019, Tommasi 2021).

Clothing production, long a site of egregious worker abuses, has provided a setting demonstrating both new leverage-driven approaches to addressing and redressing harm, and some of their limitations. Company codes of conduct including 'ethical' production commitments in textiles and apparel ranked amongst the earliest business and human rights initiatives, significantly pre-dating the GPs (Compa and Hinchcliffe Darricarrère, 1995; Methven O'Brien 2009; Beckers 2015). Though their uptake by many consumer-facing companies was relatively rapid, strong critiques emerged equally quickly. On one hand, third-party social auditors were observed to rely on a 'tickbox' approach to monitoring workplace standards for purchasers (O'Rourke 2002; Barrientos and Smith 2007, LeBaron and Lister 2015). On the other, lack of coordination amongst purchasers was seen to lead to 'audit-fatigue' for inspected businesses (Prepscius, 2012). Subsequent developments aimed to address such problems, for example, through virtual data-sharing platforms (Sedex, 2014) and capacity strengthening measures for suppliers and stakeholders (Shift, 2013b), as well as participatory and gender-sensitive approaches to monitoring code implementation (Barrientos 2005).

Notwithstanding such efforts, and global fanfare around the GPs, in 2013, over 1000 mainly female garment workers were killed and more than 2500 injured in the Savar building collapse. Various factors contributed to the "Rana Plaza" disaster, amongst them breaches of construction, health and safety regulations and labour standards by local suppliers based in the factory, under contract to well-known European and American brands, as well as defective inspection arrangements and social audits on the part of purchasers. Albeit these problems, as well as a broader context of exploitation and marginalisation of female garment workers in Bangladesh, were widely documented (Alam and Blanch, 2011) and had led to other workplace disasters before (Absar, 2002; Foxvog *et al.*, 2013), the Rana Plaza catastrophe, because of its horrendous scale, attracted public outrage, and triggered a significant multi-actor mobilisation.

Brands were convened by the ILO and global unions to coordinate an *Arrangement* for the payment of compensation to workers (ILO, 2013). In May 2013, within a few weeks of the tragedy, brands and retailers entered into a 5-year binding agreement with Bangladeshi and global trade unions. The *Accord on Fire and Building Safety in Bangladesh* committed more than 150 companies to collaborative efforts to ensure safety in almost half of the country's garment factories, through measures such as independent inspections by trained fire and building safety experts; public reporting; mandatory repairs and renovations to be financed by brands; a central role for workers and unions in both oversight and implementation; supplier contracts with sufficient financing; and adequate pricing and worker training (*Accord on Fire and Building Safety in Bangladesh*, 2013; Foxvog et al., 2013; Kenner and Peake, 2017; Rahim, 2020). Other international organisations sought to support these efforts (OECD, 2014a). Still, various companies refused to sign the *Accord*, opting instead for non-binding commitments to improved factory safety. Moreover, the *Rana Plaza Donor's Trust Fund*, set up under the *Accord* had in 2014 received only half the US\$40million needed to compensate workers or their families, while only half the companies associated with factories in the collapsed building had contributed to the fund at all (Clean Clothes Campaign, n.d.).

What caused the Rana Plaza disaster, and what steps can be taken to avoid similar events in future? These are pressing questions for business and human rights, but their answers are complex. Clearly the GPs were not, and cannot ever be, a silver bullet solution, in this regard. Some research suggests that codes of conduct and multi-stakeholder initiatives built around them may have protective effects under certain circumstances (Egels-Zandén and Lindholm 2015; Baumann-Pauly et al 2015). To the extent the GPs encourage and provide a benchmark standard for scrutinising such efforts, this appears positive. Yet overall, a global market environment that is intensely competitive, especially for

suppliers, presents systemic if not structural challenges for upholding labour standards (Marslev, 2019) while local social, political and economic factors that global initiatives may struggle to influence also undermine respect for workers' human rights (Siddiqui and Uddin, 2016). Further, while supply relationships in some sectors are relatively static and concentrated (Guglielmo, 2013) contractual networks are as dynamic as they are vast for many companies. Commodities can present their own distinct, if not unsurmountable, challenges, for instance, in terms of traceability (United Nations Global Compact and BSR, 2014). Still, if concerted multi-actor, multi-level responses as seen for Rana Plaza embody some positive potentials in governance terms, the fact remains that it is 'too little, too late' for the families of those who were injured or perished. Inevitably, as long as such man-made disasters continue, the capacity for ethical action by large, resource-rich corporations and their investors will be doubted, while calls for 'hard' international human rights law interventions to hold buyers responsible for the delinquencies of their supply chain will be heard, whether apt to address root causes of abuses, or not (LeBaron and Lister 2021, cf. Mantouvalou, 2018; Backer 2017).

# 3.3. Corporate human rights reporting

In parallel to ethical investment efforts undertaken for their own sake recent years have also witnessed widening belief in the 'materiality' of social and sustainability issues to fiduciary duties and investor returns (Hopwood et al., 2010; Unerman and Zappettini, 2014; McCorquodale, Nolan, 2021), as foreseen by the idea of the 'triple bottom line' (Elkington 1994). Corporate 'non-financial' reporting, as a device through which companies analyse, document and deliver a public account of their sustainability performance has in tandem become increasingly prominent, to the extent some have pointed to a "disclosure revolution" (Cooper and Owen, 2007; Islam and McPhail, 2011; Hohnen, 2012).

Connecting with these broader trends, the final step called foreseen by the GPs' due diligence process is for businesses to "communicate" on how they address, not just sustainability issues generally, but human rights impacts specifically (UNHRC, 2011, GPs 20-21). This may be done through formal and informal public reporting, in-person meetings, online dialogues, or consultations with rights-holders. Information provided by companies should be: (i) published in a format, and with a frequency, matching the scope and severity of impacts; (ii) accessible to intended audiences; (ii) sufficient to permit evaluation of the adequacy of company responses; (iii) designed not to pose risks to rights-holders or others such as human rights defenders, journalists, local public officials or company personnel, or to breach legitimate commercial confidentiality requirements. Businesses whose operations or operating contexts pose risks of severe human rights impacts are expected to report formally (GP21).

Since 2011, many regulators have advanced new requirements on companies to report on their human rights impacts. For example, in France, 'soft' (comply or explain) obligations mandating corporate social reporting by publicly-listed companies were first introduced in 2001. Matters to be addressed by such reports included employee contracts, working hours, pay, industrial relations, health and safety, disability policies, community relations and environmental reporting. In 2012, the reporting duty was strengthened and extended to human rights, with explicit reference made to the GPs (République Française, 2012; French Ministry of Foreign Affairs and International Department, n.d.). Also in 2012, a 'soft' non-financial reporting duty for the largest 1100 companies and Danish state-owned enterprises was extended to human rights and climate change (Danish Government, 2011, 2012). In 2013, Norway enacted rules requiring companies to report on steps to integrate human rights into their strategies (UNEP, KPMG, 2013) and the UK introduced a requirement on all companies to prepare a "Strategic Report" amongst others disclosing their "position, performance

and future direction" relating to human rights (United Kingdom, 2013). In the same period, the US enacted measures obliging companies to publish information on policies and processes concerning risks to human rights, in connection with new investments in Myanmar (U.S. Treasury Department Office for Assets Control (2013); U.S. Department of State, 2013).

At EU level, the 2014 Non-Financial Reporting Directive adopted a "report or explain" approach not dissimilar to the French and Danish models described above (EU, 2014; European Coalition for Corporate Justice, 2014). Under the 2014 Directive, "public interest enterprises" with more than 500 employees needed to report annually on their principal risks on human rights, environment and social impacts linked to their operations, relationships, products and services, as well as bribery and diversity. In turn this entailed providing information on relevant policies, due diligence procedures for identifying, preventing and mitigating risks, and significant incidents occurring during the reporting period. Voluntary frameworks on corporate human rights reporting were newly established (UN Guiding Principles Reporting Framework) or, where they already existed, aligned to the GPs (Global Reporting Initiative (GRI) (GRI, 2013; GRI, 2022) and extended, in tandem with the above legislative developments, in order to support companies in meeting their requirements (Harper Ho and Park 2019).

Whereas the EU Directive was welcomed as a step towards greater corporate accountability (European Coalition for Corporate Justice, 2014), it was also criticised *inter alia* for its narrow personal scope (it covered only 6,000 of 42,000 large companies incorporated in the EU); its potentially wide-ranging exemptions in relation to information that should be disclosed; weak provisions on supply chain reporting which was required only "when relevant and appropriate"; and for failure to provide for monitoring or mechanisms to sanction company defaults on fulfilling reporting duties: auditors needed only to indicate whether non-financial information was provided, or not (European Coalition for Corporate Justice, 2014).

Given such limitations, along with more general critiques of business efforts to adhere to human rights as mentioned above, and political concerns to accelerate and advance 'green' investment activity, a new EU Corporate Sustainability Reporting Directive was proposed in 2021. Its intention is to provide 'the foundation of a consistent flow of sustainability information through the financial value chain', from companies undertaking business activities to 'analysts in banks, insurance companies, asset management companies or credit rating agencies, for end investors and for non-governmental organisations and other stakeholders that wish to better hold companies to account for their social and environmental impacts'. Its personal scope is broader, applying to all large companies (approximately 49,000). Instead of the non-binding guidance provided for under the 2014 Directive, mandatory EU sustainability reporting standards will follow via delegated acts (European Commission 2021, Baumüller & Grbenic, 2021, Methven O'Brien, 2021).

Such strengthening measures may serve to address shortcomings of the EU's 2014 reporting legislation. Still, scholars consistently question the value of corporate sustainability reporting as an accountability mechanism (Narine, 2015; McCorquodale and Nolan, 2021; Favotto and Kollman 2022). What if, for instance, the businesses that most need to report on human rights are reluctant to do so, given commercial sensitivities, potential legal liability, the likelihood of reputational damage or internal capacity constraints? (Marquis, Toffel and Zhou 2011; Preuss and Brown, 2012, Lim 2017). One survey of corporate reports undertaken for the GRI and the UNGC identified some innovative approaches by companies to human rights reporting but concluded that, overall, it was weak in terms of balance, completeness, and inclusion of the most relevant issues (Umlas, 2009).

Other studies point to the instrumentalization of reporting to serve corporate ends (Diouf and Boiral 2017) in conflict with sustainability goals (Cho et al 2015). Activists and scholars likewise criticise published reports as exercises in "green-" or "blue-washing", based on companies' selective approach to the information that is communicated (Adams, 2004; Horiuchi and Schuchard, 2009; Gray, 2010; Boesso *et al.*, 2013).

If the development of universal human rights indicators seems crucial for comparability across company reports, their potential for irrelevance, linkage to perverse outcomes and selectivity is emphasised by others (Engel Merry, de Felice, 2015). At any rate, there is no generally accepted definition of what the 'S' in the ESG should incorporate or how to measure it, so the information generated may be of limited utility in informing due diligence measures (O'Connor and Labowitz, 2017). Geographically, there are also gaps. Although China (Wu and Habek, 2021; Notice for Better Preparing 2008 Annual Reports of Listed Companies, art. 11; CERDS, 2022) and India (Government of India, 2018; SEBI, 2021) have taken steps to require companies to disclose CSR-related information, most such measures do not directly refer to GPs-based due diligence or human rights, exacerbating problems of comparability, while reporting and disclosure requirements are completely lacking in many other jurisdictions.

In addition, civil society actors and researchers may not have the resources or interest to sustain the role, sometimes foreseen for them, of monitoring the content and quality of non-financial reports (The Landmark Project, 2012) even if have some have risen to the task (Martin-Ortega 2017; Stevenson and Cole 2018; Alliance for Corporate Transparency, 2021; Schaper and Pollach 2021). Still, the volume of material requiring review and assessment is astronomical: companies participating in the UN Global Compact (UNGC) doubled from 2016 to 2022, to 16,169 across 161 countries, issuing 92,867 public reports (UNGC2022), for example. Besides, horizontal reporting requirements typically fall short of requiring supply chain transparency, which is viewed as essential to addressing abuses in some contexts (Smit et al, 2020), as exemplified by the Bangladesh Accord which requires signatories to disclose details of their supplier factories (Accord on Fire and Building Safety in Bangladesh, n.d.).

The GPs maintain that "independent verification of human rights reporting can strengthen its content and credibility" (UNHRC, GP21) and 'professional' assurance of corporate sustainability reports has been advanced as one solution to such dilemma (SHIFT, n.d.). Proposed new laws further foresee a role for auditors in this regard (European Commission, 2021, art.1(10), art. 3(12)). But the quality and reliability of assurance has also been consistently questioned (Fonseca, 2010; O'Dwyer et al., 2011; Kaspersen and Johansen, 2014) with many human rights advocates insisting, for example, that only worker-based approaches to monitoring can be trusted (Electronics Watch, 2014; Workers' Rights Consortium, 2016). Ultimately, in this area, it seems likely that a more potent mix of mandatory disclosure rules, participatory monitoring, and continuing, enhanced investor and civil society scrutiny of company information will be needed if reporting's potential as a lever to improve corporate sustainability and business respect for human rights is to be realised (Methven O'Brien and Martin Ortega 2020). Moreover, despite the overlaps, gaps and complexity entailed, there seems no way to avoid the co-existence of a multitude of public and private governance approaches in the domain of corporate human rights disclosure, their oversight and evaluation (Harper Ho and Park 2019).

# 4. Evaluating the corporate responsibility to respect human rights and due diligence: forward momentum but unclear destination?

Uptake of due diligence, and realisation of the corporate responsibility to respect human rights amongst businesses remains fitful. Still in 2020 almost half (46.2%) of the world's biggest companies did not demonstrate any measures to identify or mitigate human rights issues in their supply chains, for example (Corporate Human Rights Benchmark, World Benchmarking Alliance 2020). Amongst 1,000 of the world's most influential companies examined in 2022, over three quarters (78 per cent) scored zero per cent on human rights due diligence indicators, with only 7 per cent of those assessed meeting all requirements (World Benchmarking Alliance, 2022). Only 5% could provide a clear example of specific conclusions and actions relating to a salient human rights issue, as a result of their assessment process, in the previous three years. According to another 2022 analysis, just 15% of companies include workers in due diligence processes and only 1 in 5 companies provide evidence of responsible purchasing practices (Human Rights Resource Center, Know the Chain, 2022).

New legal norms based on the GPs, including its elements of due diligence and human rights reporting, that might extend and accelerate changes in corporate practice therefore seem needed. On the other hand, evidence on the effectiveness of existing measures in preventing or remediating business-related abuses points in various directions (European Commission et al, 2020; McCorquodale, Nolan 2022; Methven O'Brien, 2022). Analyses undertaken regarding France's 2014 Loi de Vigilance, for example (EDH and BL Evolution, 2018; Shift 2019, Sherpa et al 2019; EDH 2020; EY 2019; Development International e.V 2020; Schilling Vacaflor 2020; Duthilleul, De Jouvenel, 2020; Sherpa, CCFD Terre Solidaire, 2021; Brabant, Savourey, 2021) have found, inter alia, that companies' vigilance plans are not transparent in terms of "which risks have been identified by the businesses [...] and even less how companies respond to them" (Sherpa, 2019, p.10); and that a substantial minority of companies (44 of 263 French companies subject to the law) did not publish a plan at all. Though 6 companies have been summoned to court or have received formal notices from organisations and unions (Sherpa, CCFD Terre Solidaire 2021), none has yet yielded a final outcome favouring victims, while alert and complaint mechanisms have been spotlighted as inaccessible for users and offering uncertain protection for victims (Schilling Vacaflor, 2020). On the other hand, an assessment of human rights reporting by the 20 largest French companies showed that 55% of companies seemed slightly to improve their human rights disclosure under the influence of new legal requirements (SHIFT, 2019).

Studies similarly present a mixed picture of the impacts of thematic or value-chain specific due diligence laws, as illustrated by the case of conflict minerals. An analysis undertaken in 2013 found only 4 per cent of 330 companies surveyed were voluntarily preparing a public report on how they identified and addressed the risk of funding conflict or abuses in their supply chains (European Commission, 2014b). Against such a benchmark, new regulations appeared likely to prompt positive change in companies' sourcing practices. Yet amongst later assessments of the impacts of new EU conflict mineral rules situation at mines in conflict-affected areas, a number concerningly detected increases in the exposure to conflict and poverty amongst communities the law was aimed to support (Enough Project, 2015; cf. Wolfe, 2015; Diemel, Hilhorst 2018; Wakenge et al, 2021; Stoop et al, 2018).

If the efficacy of due diligence-based import bans (Pietropaoli, Johnstone, Balch, 2021) and related measures (Green, 2021) in combating forced labour and human trafficking has also been queried, 'unilateral' regulatory measures by states in the global North have attracted more general and

fundamental critiques. Whereas their consistency with the right to self-determination (Okowa, 2020) as well as their democratic legitimacy, have been challenged (Knapp, 2022; Ferrarini, 2022), concerns that such laws merely distract from measures that would more directly empower marginalised rightsholders, such as increasing wages, have also been voiced (Lebaron, 2021; Nolan 2022).

A further risk of due diligence laws is that they may result in the imposition on companies of generic due diligence requirements that miss the GPs' purpose of inspiring externally structured but self-authored and context-sensitive transformations in corporate ideals and behaviour. If the challenge is to "create the structural premises for decentralized integration of society by supporting integrative mechanisms within autonomous social subsystems" (Teubner, 1983, p, 255), then balancing soft and hard laws is essential (Ruggie 2020; Choudhury 2018). On this view, what is needed is establishing linking mechanisms between formal legal frameworks and soft law and governance mechanisms deployed by private, social as well as state economic actors (Methven O'Brien and Martin-Ortega 2020, Backer 2020), rather than totally supplanting the former by the latter (Ruggie, 2020). It would hence also seem premature, in an international business and human rights treaty, to nominate specific legislative models for adoption, when strong indications of success for any one regulatory design are currently lacking, and a more open-ended regulatory design, as might be embodied in a framework-style instrument, might on this basis appear preferable (Methven O'Brien 2020, 2022).

Proposals for new due diligence laws nevertheless continue to emerge apace (Parliament of Canada, 2020 proposal of Modern Slavery Act; New Zealand Ministry of Business, Innovation and Employment 2022, Consultation on Modern Slavery and Worker Exploitation). Their risks and limitations, as discussed above, do not entail that such initiatives should be abandoned. On the contrary, it seems plausible that statutory due diligence duties, if backed by adequate support for labour rights and participation, remedies for victims, penalties and administrative enforcement may stimulate positive changes to business practices (Methven O' Brien, Martin-Ortega, 2020). Given the continuing scale and urgency of business-related human rights abuses, what is however in addition needed is to intensify inter-disciplinary scientific as well as social and political scrutiny of the impacts, design and coordination of due diligence laws, supporting and off-set reforms to labour, trade and investment regimes, and development assistance, to provide a better-informed basis for regulators embarking on and charged to oversee and enforce legislative initiatives.

### 5. Conclusion

The GPs assert the corporate responsibility to respect human rights as a free-standing, universally-applicable minimum standard of business conduct, driven by social expectations, but implied by international law. Though it may arguably be criticised for being a legal "fudge", its hybrid status can perhaps be understood, and forgiven, as a necessary compromise: it recognises, simultaneously, the enduring role of states as *de jure* duty bearers under human rights standards, but also the ethically unacceptable limitations of the latter, given the still state-centric structure of international law, and the harmful externalities affecting people and planet of corporate exploitation of globalisation's governance "gaps".

As disclosed in this chapter, the corporate responsibility to respect human rights embodies a certain culmination of developments in the sphere of corporate accountability for human rights over recent decades, while it has also served as a catalyst for its further evolution. Based on the UN Framework's second 'Pillar', diverse innovations in regulatory praxis have unfolded, across government and the corporate sector, civil society, labour unions and others, in the areas of human rights due diligence,

human rights impact assessment, measures to address corporate impacts on human rights as well as corporate human rights reporting.

Yet overall, change on the ground has remained slow and partial and results achieved to date cannot be deemed satisfactory: severe business-related human rights abuses remain endemic in many industry sectors and countries, while aggregate impacts threaten communities, climate, biodiversity and, for many, existence itself. If it is too soon to draw firm conclusions regarding the "true" impact of the responsibility to respect and due diligence, what can be counted as certain are the pressing needs for constant critique, careful scrutiny of claims of "progress", and a watchful eye on whether current approaches to regulation of the human rights impacts of business are or can be fit for purpose, within current legal, developmental and social parameters.

## References

- Aaronson, S.A. and Higham, I. (2013), ""Re-righting business": John Ruggie and the struggle to develop international human rights standards for transnational firms", *Human Rights Quarterly*, Vol. 35 No. 2, pp. 333-364.
- Abrahams, D. and Wyss, Y. (2010), "Guide to human rights impact assessment and management (HRIAM)", *International Business Leaders Forum and International Finance Corporation*, available at:
  - http://www.ifc.org/wps/wcm/connect/8ecd35004c0cb230884bc9ec6f601fe4/hriam-guide-092011.pdf?MOD=AJPERES(accessed 20 July 2022).
- Absar, S.S. (2002), "Women garment workers in Bangladesh", *Economic and Political Weekly*, Vol. 37 No. 29, pp. 3012-3016.
- Accord on Fire and Building Safety in Bangladesh (n.d.), "The Bangladesh Accord on Fire and Building Safety", available at: http://bangladeshaccord.org/ (accessed 20 July 2022).
- Adams, C.A. (2004), "The ethical, social and environmental reporting-performance portrayal gap", *Accounting, Auditing & Accountability Journal*, Vol. 17 No. 5, pp. 731-757.
- Aizawa M., Bradlow D., Wachenfeld M. (2018) "International financial regulatory standards and human rights: connecting the dots". *Manchester Journal of International Economic Law* 15(1):2–44.
- Alam, K. and Blanch, L. (2011), "Stitched up. Women workers in the Bangladeshi garment sector", *War on Want*, available at:
  - http://media.waronwant.org/sites/default/files/Stitched%20Up.pdf (accessed 20 July 2022).
- Alliance for Corporate Transparency (2021), "New data on companies' climate and human rights disclosures: lessons for the EU reform", available at:https://www.allianceforcorporatetransparency.org/news/new-data-on-companies-climate
  - and-human-rights-disclosures-lessons-for-the-eu-reform.html (accessed 20 July 2022).
- Amnesty International(2019), "Outsourcing Responsibility Human Rights Policies In The Defence Sector", available at <a href="https://www.amnesty.de/sites/default/files/2019-09/Amnesty-Bericht-">https://www.amnesty.de/sites/default/files/2019-09/Amnesty-Bericht-</a>
- Outsourcing-Responsibility-Ruestungsunternehmen-September 2019. PDF (accessed 20 July 2022).
- Australian Government(2012), Illegal Logging Prohibition Act, available at
  - https://www.agriculture.gov.au/agriculture-land/forestry/policies/illegal-logging (accessed 20 July 2022).
- Australian Government (2018), The Modern Slavery Act, available at:
  - https://www.legislation.gov.au/Details/C2018A00153 (accessed 20 July 2022).
- BankTrack and the Berne Declaration, (2012), "On The Thun Group Paper On

- Banks And Human Rights", available at <a href="https://www.banktrack.org/download/banktrack">https://www.banktrack.org/download/banktrack</a> on the thun group paper on banks and human rights/131129 thun group paper final.pdf (accessed 20 July 2022).
- Barrientos, S., and Smith, S. (2007), "Do workers benefit from ethical trade? Assessing codes of labour practice in global production systems", *Third World Quarterly*, 28(4), 713-729. https://doi.org/10.1080/01436590701336580
- Baumann-Pauly, D., Nolan, J, Van Heerden, A. and Samway, M., (2015) "Industry-Specific Multi-Stakeholder Initiatives that Govern Corporate Human Rights Standards Legitimacy Assessments of the Fair Labor Association and the Global Network Initiative", *UNSW Law Research Paper* No. 2015-12, Available at <a href="http://dx.doi.org/10.2139/ssrn.2576217">http://dx.doi.org/10.2139/ssrn.2576217</a>
- Baker, A. (2017), "The Corporate Social Responsibilities of Financial Institutions for the Conduct oftheir Borrowers: The View from International Law and Standards", 21 *Lewis & Clark L. Rev.* 881
- Backer. L.C. (2017), "A Lex Mercatoria for Corporate Social Responsibility Codes Without the State? A Critique of Legalization Within the State Under the Premises of Globalization", *Indiana Journal of Global Legal Studies*, 24(1), 115–146. https://doi.org/10.2979/indjglolegstu.24.1.0115
- Backer, A. L. (2020), "Human Rights Responsibilities of State-Owned Enterprises", in in Deva S. and Birchall D. (eds.), 2020 *Research Handbook on Human Rights and Business*, Cambridge University Press, Cambridge, pp. 221-242Baumüller, J. and Grbenic, S. (2021) "Moving from non-financial to sustainability
- reporting: analyzing the EU Commission's proposal for a Corporate Sustainability Reporting Directive (CSRD)". Facta Universitatis, Series: Economics and Organization, Vol. 18, No 4, Special Issue, 2021, pp. 369-381 available at https://doi.org/10.22190/FUEO210817026B (accessed 20 July 2022).(
- Beckers, A. (2015), Enforcing Corporate Social Responsibility Codes: On Global Self-Regulation And National Private Law, Oxford: Hart Publishing, Oxford.
- British Institute of International and Comparative Law (BIICL), Norton Rose Fulbright, (2017), "Making sense of managing human rights issues in supply chains". Available at <a href="https://www.biicl.org/newsitems/6292/biicl-and-norton-rose-fulbright-publish-supply-chain-human-rights-due-diligence-study">https://www.biicl.org/newsitems/6292/biicl-and-norton-rose-fulbright-publish-supply-chain-human-rights-due-diligence-study</a> (accessed 20 July 2022).
- Bilchitz, D. (2013), "A chasm between 'is' and 'ought'? A critique of the normative foundations of the SRSG's Framework and the Guiding Principles", in Deva, S. and Bilchitz, D. (eds.), *Human Rights Obligations of Business Beyond the Corporate Responsibility to Respect*, Cambridge University Press, Cambridge, pp. 107-137.
- Bonnitcha, J. and McCorquodale, R. (2017a) "The Concept of 'Due Diligence" in the UN Guiding Principles on Business and Human Rights, *European Journal of International Law*, 28(3):899-919.
- Bonnitcha, J., and McCorquodale, R. (2017b), "The concept of "Due Diligence" in the UN Guiding Principles on Business and Human Rights: A rejoinder to John Gerard Ruggie and John F. Sherman, III' European Journal of International Law, 28, p. 929.
- Boesso, G., Kumar, K., and Michelon, G. (2013), "Descriptive, instrumental and strategic approaches to corporate social responsibility: Do they drive the financial performance of companies differently?" *Accounting, Auditing & Accountability Journal*, Vol. 26 No. 3, pp. 399-422.
- Botta, G.(2022), "Unpacking the Potentials of a Framework Agreement on Business and Human

- Rights: An Opportunity to Transcend the Hard and Soft-Law Dichotomy for More Policy Coherence", Völkerrechtsblog, doi: 10.17176/20220620-153018-0.
- Bright, C. and Buhmann, K.(2021), "Risk-Based Due Diligence, Climate Change, Human Rights and
  - the Just Transition", *Sustainability* 13, no. 18: 10454. https://doi.org/10.3390/su131810454
- Bueno, N., and Kaufmann, C. (2021), "The Swiss Human Rights Due Diligence Legislation: Between Law and Politics", *Business and Human Rights Journal*, 6(3), 542-549. doi:10.1017/bhj.2021.42
- Buhmann, K. (2020), "Meaningful Stakeholder Engagement as an Aspect of Risk-based Due Diligence between the Economy, Politics and Law: The Constitutive Role of the Business and Human Rights Regime" in Johnstone, R.L., and Hansen, A.M. (Eds.), *Regulation of Extractive Industries: Community Engagement in the Arctic* Routledge, London, pp. 79-98.
- Bundestag(2021), Gesetz über die unternehmerischen Sorgfaltspflichten in Lieferketten available at
  - https://www.bgbl.de/xaver/bgbl/text.xav?SID=&tf=xaver.component.Text\_0&tocf=&qmf=&hlf=xaver.component.Hitlist\_0&bk=bgbl&start=%2F%2F\*%5B%40node\_id%3D%27940343%27%5D&skin=pdf&tlevel=-2&nohist=1&sinst=87F4D859 (accessed 20 July 2022).
- Business and Human Rights Resource Center (2019), "List of large businesses, associations & investors with public statements & endorsements in support of mandatory due diligence regulation", available at <a href="https://www.business-humanrights.org/en/latest-news/list-of-large-businesses-associations-investors-with-public-statements-endorsements-in-support-of-mandatory-due-diligence-regulation/">https://www.business-humanrights.org/en/latest-news/list-of-large-businesses-associations-investors-with-public-statements-endorsements-in-support-of-mandatory-due-diligence-regulation/</a> (accessed 20 July 2022).
- Business and Human Rights Resource Centre (2022), "Closing the gap: Evidence for effective human rights due diligence from five years measuring company efforts to address forced labour", available at <a href="https://www.business-humanrights.org/en/from-us/briefings/closing-the-gap-evidence-for-effective-human-rights-due-diligence-from-five-years-measuring-company-efforts-to-address-forced-labour/#:~:text=KnowTheChain%20benchmarks%20how%20companies%20implement,lab
  - our% 20in% 20their% 20supply% 20chains (accessed 20 July 2022).
- Business Europe (2022), "EU companies need workable rules on corporate due diligence", available at <a href="https://www.businesseurope.eu/publications/eu-companies-need-workable-rules-corporate-due-diligence">https://www.businesseurope.eu/publications/eu-companies-need-workable-rules-corporate-due-diligence</a> (accessed 14 July 2022)
- Carrillo Santarelli, N., Arévalo-Narváez C., (2017) "The discursive use and development of the guiding principles on business and human rights in Latin America", International Law: Revista Colombiana de Derecho Internacional, núm. 30, enero-junio, pp. 61-117, Pontificia Universidad Javeriana, Bogotá, Colombia
- Cernic, J. L., and Van, H. T. (2015), *Human rights and business: Direct corporate accountability for human rights*, Wolf Legal Publishers, Oisterwijk, The Netherlands.
- China Chamber of Commerce of Metals, Minerals and Chemicals Importers and Exporters (2015), Chinese Due Diligence Guidelines for Responsible Mineral Supply Chains, available at: https://mneguidelines.oecd.org/chinese-due-diligence-guidelines-for-responsible-mineral-supply-chains.htm (accessed 20 July 2022).
- Cho C.H., Laine, M., Roberts, R.B., Rodrigue, M. (2015), "Organized hypocrisy, organizational façades, and sustainability reporting", *Accounting, Organizations and Society*, Vol. 40, (C), pp. 78-94

- Choudhury, B. (2018), "Balancing Soft And Hard Law For Business And Human Rights", *International and Comparative Law Quarterly*, 67(4), 961-986. doi:10.1017/S0020589318000155.
- Clapham, A. (1996), Human Rights in the Private Sphere. Oxford: Clarendon.
- Clean Clothes Campaign (n.d.), "Who needs to pay up?", available at: http://www.cleanclothes.org/ranaplaza/who-needs-to-pay-up (accessed 20 July 2022).
- Cohn, C. Timm, T., and York, J. (2012), "Human rights and technology sales: how corporations can avoid assisting repressive regimes", *Electronic Frontier Foundation*, available at: https://www.eff.org/files/filenode/human-rights-technology-sales.pdf (accessed 20July 2022).
  - Columbia Center on Sustainable Investment, Danish Institute for Human Rights, and Sciences Po Law School Clinic (2017), "A Collaborative Approach to Human Rights Impact Assessments", available at: <a href="https://ccsi.columbia.edu/sites/default/files/content/A-Collaborative-Approach-to-HRIAs\_Web.pdf">https://ccsi.columbia.edu/sites/default/files/content/A-Collaborative-Approach-to-HRIAs\_Web.pdf</a> (accessed 9July 2022).
- Compa, L. A., Hinchliffe-Darricarrère, T.(1995), "Private Labor Rights Enforcement Through Corporate Codes of Conduct" *Columbia Journal of Transnational Law* 33:663, pp. 665-686
- Corporate Human Rights Benchmark, World Benchmarking Alliance (2020), "Corporate Human Rights Benchmark: Key Findings Report", available at:
  <a href="https://www.worldbenchmarkingalliance.org/research/corporate-human-rights-benchmark-2020-key-fndings-report/">https://www.worldbenchmarkingalliance.org/research/corporate-human-rights-benchmark-2020-key-fndings-report/</a> (accessed July 2022).
- Corte Costitutional (2016), La Sala Quinta
  - de Revisión de la Corte Constitucional, *Acción de tutela instaurada por José Eliecer Díaz Bohórquez contra el Oleoducto Central Andino Ocensa y el ConsorciodeTierras Boyacá*, Sentencia T-732/16.
- Cooper, S. and Owen, D. (2007), "Corporate social reporting and stakeholder accountability: the missing link", *Accounting, Organizations and Society*, Vol. 32 No. 7-8, pp. 649-667.
- Council of Europe (2014), "Declaration of the Committee of Ministers on the UN Guiding Principles on Business and Human Rights", available at: <a href="https://search.coe.int/cm/Pages/result\_details.aspx?ObjectId=09000016805c6ee3">https://search.coe.int/cm/Pages/result\_details.aspx?ObjectId=09000016805c6ee3</a> (accessed 20 July 2022).
- Council of Europe (2016) Human Rights and business Recommendation CM/Rec(2016)3 of the Committee of Ministers to Member States (2016), available at <a href="https://edoc.coe.int/en/fundamental-freedoms/7302-human-rights-and-business-recommendation-cmrec20163-of-the-committee-of-ministers-to-member-states.html">https://edoc.coe.int/en/fundamental-freedoms/7302-human-rights-and-business-recommendation-cmrec20163-of-the-committee-of-ministers-to-member-states.html</a>.
- Danish Government (2012), Responsible Growth the Action Plan for Corporate Social Responsibility 2012-15, available at: http://csrgov.dk/danish\_action\_plan\_2012 (accessed 20 July 2022).
- Danish Government (2011), Corporate Social Responsibility and Reporting in Denmark: Impact of the third year subject to the legal requirements for reporting on CSR in the Danish Financial Statements Acts, available at:
  - http://csrgov.dk/studies\_impact\_of\_legal\_requirement (accessed 20 July 2022).
- Danish Institute for Human Rights (n.d.), *National Action Plans on Business & Human Rights*, available at <a href="https://globalnaps.org/">https://globalnaps.org/</a> (accessed 12 July 2022)
- Danish Institute for Human Rights (2019), *Human rights and state-investor contracts*, available at <a href="https://www.humanrights.dk/tools/human-rights-state-investor-contracts">https://www.humanrights.dk/tools/human-rights-state-investor-contracts</a> (accessed 20 July 2022).

- Danish Institute for Human Rights (2020), *Human Rights Impact Assessment Guidance and Toolbox*, available at: <a href="https://www.humanrights.dk/tools/human-rights-impact-assessment-guidance-toolbox">https://www.humanrights.dk/tools/human-rights-impact-assessment-guidance-toolbox</a> (accessed 20 July 2022).
- Danish Institute for Human Rights (2020), *Guidance on Human Rights Impact Assessment of digital activities*, available at: https://www.humanrights.dk/publications/human-rights-impact-assessment-digital-activities.
- Danish Institute for Human Rights and Nestlé (2013), "Talking the human rights walk. Nestlé's experience assessing human rights impacts in its business activities", available at: http://www.nestle.com/asset-library/documents/library/documents/corporate\_social\_responsibility/nestle-hria-white-paper.pdf (accessed 20 July 2022).
- Davies, A. (2008), "One Step Forward, Two Steps Back? The *Viking* and *Laval* Cases in the ECJ", *Industrial Law Journal*, Vol. 37(2), pp.126-148.
- Davies, R. and Vadlamannati, K. (2013), "A race to the bottom in labor standards? An empirical investigation", *Journal of Development Economics*, Vol. 103, pp.1-14.
- Debevoise & Plimpton (2021) UN Guiding Principles on Business and Human Rights at 10

  The Impact of the UNGPs on Courts and Judicial Mechanisms, available at:

  <a href="https://www.ohchr.org/sites/default/files/Documents/Issues/Business/UNGPsBHRnext10/debevoise.pdf">https://www.ohchr.org/sites/default/files/Documents/Issues/Business/UNGPsBHRnext10/debevoise.pdf</a> (accessed 20 July 2022).
- De Albuqurque, B.M., (2019), "Human Rights Due Diligence in International Law: Where Do We Go from Here?", in Pinto P., de Albuquerque, Wojtyczek K.,(eds.), 2019, *Judicial Power in a Globalized World*, available at: https://doi.org/10.1007/978-3-030-20744-1\_16 (accessed 20 July 2022).
- De Felice, D. (2015), "Challenges and opportunities in the production of business and human rights indicators to measure the corporate responsibility to respect", *Human Rights Quarterly*, Vol. 37 No. 2, pp. 511-555.
- Deva S. (2012), Regulating Corporate Human Rights Violations: Humanizing Business London; New York: Routledge, 2012, ISBN 978-0-415-66821-7, available at: https://ssrn.com/abstract=2034793Deva S., Brichall D.,(2020), *Research Handbook on Human Rights and Business*, Edward Elgar Publishing, Northampton.
- Development International e.V. (2020), "Devoir de Vigilance: Reforming Corporate Risk Engagement", available at: <a href="https://www.ipoint-systems.com/fileadmin/media/downloads/Devoir-de-Vigilance\_Loi-2017-399\_Study\_2020.pdf">https://www.ipoint-systems.com/fileadmin/media/downloads/Devoir-de-Vigilance\_Loi-2017-399\_Study\_2020.pdf</a> (accessed 20 July 2022);
- De Winter-Schmit, Slacito K., (2019), "The Need For A Multidisciplinary HRIA Team: Learning And Collaboration Across Fields Of Impact Assessment" in Gotzmann (ed.) 2019, Handbook on Human Rights Impact Assessment Elgr Editor, https://doi.org/10.4337/9781788970006.00030, pp. 319–335
- Dhanarajan, S. and Methven O'Brien, C. (2015), "Human rights and businesses: Emergence and development of the field in Asia, Europe and globally" in Asia-Europe Foundation, *Human Rights and Businesses. Proceedings of the 14<sup>th</sup> Informal Asia-Europe Meeting (ASEM) Seminar on Human Rights, Hanoi, Vietnam, 18-20 November 2014.* Singapore: Asia-Europe Foundation, 35-100, available at: http://www.asef.org/projects/themes/governance/3136-14-informal-asem-seminar-on-human-rights (accessed 20 July 2022).
- Diemel, J., Hilhorst, DJM. (2019), "Unintended consequences or ambivalent policy objectives? Conflict minerals and mining reform in the Democratic Republic of Congo", *Dev Policy Rev*. 2019; 37: 453–469,https://doi.org/10.1111/dpr.12372.

- Diouf, D., & Boiral, O. (2017), "The Quality of Sustainability Reports and Impression Management: A Stakeholder Perspective" *Accounting, Auditing & Accountability Journal*, Vol. 30, No.3).
- Duthilleul A., de Jouvenel, M. (2020), "Evaluation de la mise en oeuvre de la loi no. 2017-399 du 27 mars 2017 relative au devoir de vigilance des sociétés mères et des entreprises donneuses d'ordre", *Report to the French Minister of Economy and Foreign Affairs*, available at :https://www.economie.gouv.fr/cge/devoir-vigilances-entreprises (accessed 20 July 2022).
- Economist Intelligence Unit (EIU), (2015), "The Road from Principles To Practice Today's Challenges For Business In Respecting Human Rights", available at: <a href="https://www.universal-rights.org/urg-policy-reports/the-road-from-principles-to-practice-todays-challenges-for-business-in-respecting-human-rights/">https://www.universal-rights.org/urg-policy-reports/the-road-from-principles-to-practice-todays-challenges-for-business-in-respecting-human-rights/</a> (accessed 20 July 2022).
- Ebert, F.C., Soares Mariano Costa L., (2019), "Fighting modern slavery in Brazil: The Fazenda Brasil Verde decision of the Inter-American Court of Human Rights and current implementation challenges", Völkerrechtsblog, 16 December 2019, doi: 10.17176/20191216-180649-0.
- EDH (2019), "Application of the Law of the Corporate Duty of Vigilance, Vigilance Plans 2018–2019", available at: https://www.e-dh.org/userfiles/Etude% 20plans%20de%20vigilance%202019%20-%20VEN.pdf (accessed 20 July 2022).
- EDH and BL Evolution, (2018), Application of the Law of the Corporate Duty of Vigilance, Analysis of the First Published Plans, 1st edition, available at: <a href="https://www.e-dh.org/userfiles/Edh\_2018\_Etude\_EN\_1.pdf">https://www.e-dh.org/userfiles/Edh\_2018\_Etude\_EN\_1.pdf</a>, (accessed 20 July 2022).
- Egels-Zandén, N., Lindholm, H. (2015), "Do codes of conduct improve worker rights in supply chains? A study of Fair Wear Foundation", *Journal of Cleaner Production* 107: 31-40.
- Electronics Watch (2014), *Electronics Watch. Improving working conditions in the global electronics industry*, available at: http://electronicswatch.org/en, (accessed 20 July 2022).
- Elkington, J. (1994), "Enter the Triple Bottom Line", ES TBL 7/1, available at: <a href="http://kmhassociates.ca/resources/1/Triple%20Bottom%20Line%20a%20history%201961-2001.pdf">http://kmhassociates.ca/resources/1/Triple%20Bottom%20Line%20a%20history%201961-2001.pdf</a> (accessed 20 July 2022).
- Enough Project (2015), "Dodd-Frank 1502 Law: Impact on Conflict Minerals Trade in Congo", available at: http://www.enoughproject.org/news/dodd-frank-1502-law-impact-conflict-minerals-trade-congo (accessed 20 July 2022).
- Engle Merry, S. (2011), "Measuring the World: Indicators, Human Rights, and Global Governance", , *Corporate Lives: New Perspectives on the Social Life of the Corporate Form*: Vol. 52, No. S3, available at <a href="http://www.jstor.org/stable/10.1086/657241">http://www.jstor.org/stable/10.1086/657241</a>
  Equator Principles Association (2003), "About the Equator Principles; A financial industry benchmark for determining, assessing and managing environmental and social risk in projects", available at: <a href="https://equator-principles.com/about-the-equator-principles/">https://equator-principles.com/about-the-equator-principles/</a> (accessed 20 July 2022).
- Equator Principles (2020), "Equator Principles EP4", available at <a href="https://equator-principles.com/app/uploads/The-Equator-Principles\_EP4\_July2020.pdf">https://equator-principles\_com/app/uploads/The-Equator-Principles\_EP4\_July2020.pdf</a> (accessed 20 July 2022).
- ESCR-Net CAWG (2020), "IACHR Inputs to the Questionnaire for the thematic report on Business and Human Rights: Inter-American Standards", available at <a href="https://aida-americas.org/sites/default/files/resources\_files/Inputs\_IACHR\_Business%20and%20Human%20Rights\_ENG.pdf">https://aida-americas.org/sites/default/files/resources\_files/Inputs\_IACHR\_Business%20and%20Human%20Rights\_ENG.pdf</a> (accessed 20 July 2022).
- Essers, L. (2014), "Google, Facebook to discuss online extremism at dinner with EU officials", *PCWorld*, 8 October, available at: http://www.pcworld.com/article/2813132/google-facebook-to-discuss-online-extremism-at-dinner-with-eu-officials.html (accessed 20 July 2022).

- European Coalition for Corporate Justice (2014a), "EU Directive on the disclosure of non-financial information by certain large companies: an analysis", available at: http://www.corporatejustice.org/On-15-April-2014-the-European.html?lang=en (accessed 20 July 2022).
- European Coalition for Corporate Justice (2014b), "Non-financial reporting reform on thin ice", *European Coalition for Corporate Justice*, 27 January, available at: http://www.corporatejustice.org/Non-financial-reporting-reform-on.html?lang=en (accessed 20 July 2022).
- European Coalition for Corporate Justice (2014c), "Why is the corporate reporting reform important?", available at: http://www.corporatejustice.org/IMG/pdf/media\_briefing\_26-02-2014.pdf (accessed 20 July 2022).
- European Commission, (2011) A renewed EU strategy 2011-14 for Corporate Social Responsibility, Communication, COM (2011) 681 final.
- European Commission (2013), "An analysis of policy references made by large EU companies to internationally recognised CSR guidelines and principles", available at: http://ec.europa.eu/DocsRoom/documents/10372/attachments/1/translations/en/renditions/native (accessed 20 July 2022).
- European Commission (2014a), "Proposal for a Regulation of the European Parliament and of the Council for setting up a Union system for supply chain due diligence self-certification of responsible importers of tin, tantalum and tungsten, their ores, and gold originating in conflict-affected and high-risk areas" available at: http://eur-lex.europa.eu/legal-content/EN/TXT/?uri=COM:2014:0111:FIN, (accessed 20 July 2022).
- European Commission (2014b), "Assessment of due diligence compliance cost, benefits and related effects on the competitiveness of selected operators in relation to the responsible sourcing of selected minerals from conflict-affected areas", available at: http://bookshop.europa.eu/en/assessment-of-due-diligence-compliance-costs-benefits-and-related-effects-on-the-competitiveness-of-selected-operators-in-relation-to-the-responsible-sourcing-of-selected-minerals-from-conflicts-affected-areas-pbNG0213263/;pgid=Iq1Ekni0.1lSR0OOK4MycO9B0000kcFmP8je;sid=KK6JnQ1-C\_CJnVu8z1EXOm9bim31zUAdNtw=?CatalogCategoryID=ZjsKABstHnIAAAEjH5EY4e 5L, (accessed 20 July 2022).
- European Commission (2015), "Implementing the UN Guiding Principles on Business and Human Rights State of Play, Staff Working Document", SWD (2015) 144 final.
- European Commission (2017), "Guidelines on Non-financial Reporting (Methodology for Reporting Non-financial Information), Communication from the Commission, Official Journal of the European Union, C/2017/4234.
- European Commission (2019), "The European Green Deal, Communication, COM(2019) 640 final, 2019.
- European Commission(2018), Action Plan: Financing Sustainable Growth, Communication, COM (2018) 97 final, 2018.
- European Commission, Directorate-General for Justice and Consumers, Torres-Cortés, F., Salinier, C., Deringer, H., et al., (2020), "Study on due diligence requirements through the supply chain: final report", *Publications Office*, available at: https://data.europa.eu/doi/10.2838/39830 (accessed 20 July 2022).
- European Commission (2020), Proposal For a Regulation of The European Parliament and of the Council concerning batteries and waste batteries, repealing Directive 2006/66/EC and amending Regulation (EU) No 2019/1020", accessible at: <a href="https://eur-nttps

- <u>lex.europa.eu/resource.html?uri=cellar:4b5d88a6-3ad8-11eb-b27b-01aa75ed71a1.0001.02/DOC\_1&format=PDF</u> (accessed 20 July 2022).
- European Commission (2021), "Proposal for a Regulation of the European Parliament and of the
  - Council on the making available on the Union market as well as export from the Union of certain commodities and products associated with deforestation and forest degradation and repealing Regulation (EU) No 995/2010", accessible at: <a href="https://environment.ec.europa.eu/publications/proposal-regulation-deforestation-free-products\_en">https://environment.ec.europa.eu/publications/proposal-regulation-deforestation-free-products\_en</a> (accessed 20 July 2022).
- European Commission (2021), "Regulatory Scrutiny Board Opinion, Proposal for a Directive of the European Parliament and of the Council on Sustainable Corporate Due Diligence and amending Directive (EU) 2019/1937", SEC (2022) 95, 2021.
- European Commission (2021), "EU Taxonomy, Corporate Sustainability Reporting, Sustainability Preferences and Fiduciary Duties: Directing finance towards the European Green Deal", COM (2021) 188 final.
- European Commission (2021), "Proposal for A Directive of The European Parliament and of the Council amending Directive 2013/34/EU, Directive 2004/109/EC, Directive 2006/43/EC and Regulation (EU) No 537/2014, as regards corporate sustainability reporting", COM/2021/189 final, available at: https://eurlex.europa.eu/legal-content/EN/TXT/?uri=CELEX:52021PC0189.
- European Commission, (2022a), "Corporate Sustainability Due Diligence and amending Directive (EU) 2019/1937, Proposal for a Directive", COM(2022) 71 final.
- European Commission (2022b), "Impact Assessment Report Accompanying the document Proposal for a Directive of the European Parliament and of the Council on Corporate Sustainability Due Diligence and amending Directive (EU) 2019/1937", Staff Working Document, SWD(2022) 42 final.
- European Union (2010), "Regulation 995/2010 of the European Parliament and the Council of 20 October 2010 Laying down the Obligations of Operators who Place Timber and Timber Products on the Market", OJ L295/23, *Official Journal of the European Union*.
- European Union(2014), "Directive 2014/95/EU of the European Parliament and of the Council of 22
  - October 2014 amending Directive 2013/34/EU as regards Disclosure of Non-financial and Diversity Information by Certain Large Undertakings and Groups", OJ L 330, Official Journal of the European Union, 15 November 2014.
- European Union (2017), "Regulation 2017/821 of the European Parliament and of the Council of 17
  - May 2017 Laying down Supply Chain Due Diligence Obligations for Union Importers of Tin, Tantalum and Tungsten, Their Ores, and Gold, Originating from Conflict-affected and High-risks Areas", OJ L130/1, *Official Journal of the European* Union.
- European Union (2019), "Regulation (EU) 2019/2088 of the European Parliament and of the Council of 27 November 2019 on Sustainability-related Disclosures in the Financial Services Sector", OJ L317/1, *Official Journal of the European Union*.
- European Union Agency for Fundamental Rights (FRA) (2020), "Business and human rights access to remedy", accessible at: <a href="https://fra.europa.eu/en/publication/2020/business-human-rights-remedies#publication-tab-1">https://fra.europa.eu/en/publication/2020/business-human-rights-remedies#publication-tab-1</a> (accessed 20 July 2022).
- EY (2019), "Devoir de vigilance: analyse de la deuxième année de publication", available at :

- https://www.ey.com/Publication/vwLUAssets/ey-analyse-des-premiers-plans-de-vigilance-du-sbf-120/\$File/ey-analysedes-premiers-plans-de-vigilance-du-sbf-120.pdf (accessed 20 July 2022).
- Favotto, A. and Kollman, K. (2022), "When rights enter the CSR Field: British firms' engagement with human rights and the UN Guiding Principles", *Human Rights Review*, 23(1), pp. 21-40. doi: 10.1007/s12142-020-00614-w.
  - Ferrarini, G. (2022), "Corporate Sustainability Due Diligence and the Shifting Balance between Soft Law and Hard Law in the EU", *Oxford Business Law Blog*, available at <a href="https://www.law.ox.ac.uk/business-law-blog/blog/2022/04/corporate-sustainability-due-diligence-and-shifting-balance-between">https://www.law.ox.ac.uk/business-law-blog/blog/2022/04/corporate-sustainability-due-diligence-and-shifting-balance-between</a> (accessed 20 July 2022).
- Finance Against Trafficking (n.d.), "About", available at: http://financeagainsttrafficking.org/(accessed 20 July 2022).
- Ford, J. (2015), *Business and Human Rights. Bridging the Governance Gap. International Law Programme Research Paper.* London: Royal Institute for International Affairs, available at: https://www.chathamhouse.org/sites/files/chathamhouse/field/field\_document/20150922Bus inessHumanRightsFordV2.pdf (accessed 20 July 2022).
- Ford, J. and Methven O'Brien, C., (2017) "Empty Rituals or Workable Models? Towards a Business and Human Rights Treaty", *University of New South Wales Law Journal*, Vol.40 No.3, pp. 1223-1248.
- Foxvog, L. et al. (2013), "Still Waiting: Six months after history's deadliest apparel industry disaster, workers continue to fight for reparations", *Clean Clothes Campaign and International Labor Rights Forum*, available at: http://digitalcommons.ilr.cornell.edu/cgi/viewcontent.cgi?article=2840&context=globaldocs (accessed 20 July 2022).).
- Frediani, C. (2014), "A global campaign to monitor the 'digital weapons' trade", *TechPresident*,, available at: http://techpresident.com/news/wegov/24901/curtailing-international-surveillance-tech-trade (accessed 20 July 2022).
- French Ministry of Foreign Affairs and International Department (n.d.), Extra financial reporting made mandatory for large companies in a view of a standardization of European standards, 2016, available at: http://www.diplomatie.gouv.fr/en/french-foreign-policy/economic-diplomacy-foreign-trade/corporate-social-responsibility/france-s-domestic-csr-policy/article/extra-financial-reporting-made (accessed 20 July 2022).
- Gap Inc. (/2012), "Monitoring and remediation", available at: http://www.gapinc.com/content/csr/html/human-rights/monitoring-and-remediation.html (accessed 20 July 2022).
- Global Reporting Initiative (2013), "The external assurance of sustainability reporting", available at: <a href="https://www.globalreporting.org/resourcelibrary/GRI-Assurance.pdf">https://www.globalreporting.org/resourcelibrary/GRI-Assurance.pdf</a> (accessed 20 July 2022).
- Global Reporting Initiative (2021), "A Short Introduction to the GRI Standards", available at: <a href="https://www.globalreporting.org/media/wtaf14tw/a-short-introduction-to-the-gristandards.pdf">https://www.globalreporting.org/media/wtaf14tw/a-short-introduction-to-the-gristandards.pdf</a>, (accessed 20 July 2022).
- Global Reporting Initiative, (2022), "Better human rights reporting needed now, but how?", available at
- https://www.globalreporting.org/media/1fjcdoei/gri-perspective human rightsreporting.pdf, (accessed 20 July 2022).
- Global Witness (2014), "Seeing the light. Responsible mineral sourcing from the DRC", available at:
  - http://www.globalwitness.org/sites/default/files/Seeing%20the%20Light%20April%202014. pdf (accessed 20 July 2022).

- Goldcorp (2010), "Human rights assessment of Goldcorp's Marline Mine", available at: http://csr.goldcorp.com/2011/docs/2010\_human\_full\_en.pdf (accessed 20 July 2022).
- Goldsmith, J., & Posner, E. (2005), *The Limits of International Law*. Oxford University Press Hathaway.
- Gonza, A.(2016), "Integrating Business and Human Rights in the Inter-American Human Rights System" *Business and Human Rights Journal*, Vol. 1 No. 2, 357.
- Gothberg, P. (2019), "Public procurement and human rights in the healthcare sector: the Swedish county councils collaborative model", in Methven O'Brien, Martin Ortega (Eds.), 2019 *Public Procurement and Human Rights*, Edward Elgar Publishing, pp. 165–179. https://doi.org/10.4337/9781788116312.00020
- Gotzmann, N., (2019), *Handbook on Human Rights Impact Assessment*, Edward Elgar Publishing, Social and Political Science
- Government of India, Ministry of Corporate Affairs (2018), "National Guidelines on Responsible Business Conduct", available at: <a href="https://www.mca.gov.in/Ministry/pdf/NationalGuildeline\_15032019.pdf">https://www.mca.gov.in/Ministry/pdf/NationalGuildeline\_15032019.pdf</a> (accessed 20 July 2022).
- Gray, R. (2010), "Is accounting for sustainability actually accounting for sustainability...and how would we know? An exploration of narratives of organisations and the planet", *Accounting, Organizations and Society*, Vol. 35 No. 1, pp. 47-62.
- Green, J. (2021), "Closing the Accountability Gap in Corporate Supply Chains for Violations of the Trafficking Victims Protection Act", *Business and Human Rights Journal*, Vol. 6 No. 3, pp. 449-489. doi:10.1017/bhj.2020.28
- Guglielmo, C. (2013), "Apple's supplier labor practices in China scrutinized after Foxconn, Pegatron reviews", *Forbes Technology*, available at: http://www.forbes.com/sites/connieguglielmo/2013/12/12/apples-labor-practices-in-china-scrutinized-after-foxconn-pegatron-reviewed/ (accessed 20 July 2022).
- Gunningham, N., Kagan, R.A., and Thornton, D. (2003), *Shades of Green: Business, Regulation and Environment*, Stanford University Press, Stanford.
- Hamm, B., Schax, A., and Scheper, C. (2013), "Human rights impact assessment of the Tampakan copper-gold project, Mindanao, Philippines", *Institute for Development and Peace, University of Duisburg-Essen*, available at: http://www.misereor.org/fileadmin/redaktion/HRIA\_Human\_Rights\_Impact\_Assessment\_T ampakan\_Copper-Gold\_Project\_August2013.pdf (accessed 20 July 2022).
- Harper Ho, V.E., and Park, S.(2019), "ESG Disclosure in Comparative Perspective: Optimizing Private Ordering in Public Reporting", *University of Pennsylvania Journal of International Law*, Vol. 41, No. 2.
- Hathaway, O. (2002), "Do Human Rights Treaties Make a Difference?" *Yale Law Journal*, Vol. 111, 2002, Boston Univ. School of Law Working Paper No. 02-03, available at: <a href="https://ssrn.com/abstract=311359">https://ssrn.com/abstract=311359</a> (accessed 20 July 2022).
- Hathaway, O. (2007), "Why Do Countries Commit to Human Rights Treaties?", the *Journal of Conflict Resolution*, Vol. 51, No. 4, pp. 588–621.
- Hiss, S. (2013), "The Politics of the Financialization of Sustainability". *Competition & Change*, Vol.17, pp. 234-247. Available at <a href="https://www.researchgate.net/publication/272250694\_The\_Politics\_of\_the\_Financialization\_of\_Sustainability">https://www.researchgate.net/publication/272250694\_The\_Politics\_of\_the\_Financialization\_of\_Sustainability</a> (accessed 20 July 2022).
- Hohnen, P. (2012), "The future of sustainability reporting", *EEDP Programme Paper 2012/02*, Chatham House, London, available at:

- http://www.chathamhouse.org/sites/files/chathamhouse/public/Research/Energy,%20Environment%20and%20Development/0112pp hohnen.pdf (accessed 20 July 2022).
- Hopwood, A. Unerman, J., and Fries, J. (2010), *Accounting for Sustainability: Practical Insights*, Earthscan, London.
- Horiuchi, R., Schuchard, R., Shea, L., and Townsend, S. (2009), "Understanding and preventing greenwash: a business guide", *BSR and Futerra*, available at: http://www.bsr.org/reports/Understanding\_\_Preventing\_Greenwash.pdf ( (accessed 20 July 2022).
- Human Rights Resource Center, Know the Chain, (2022), "Closing the gap: Evidence for effective human rights due diligence from five years measuring company efforts to address forced labour" available at <a href="https://www.business-humanrights.org/en/from-us/briefings/closing-the-gap-evidence-for-effective-human-rights-due-diligence-from-five-years-measuring-company-efforts-to-address-forced-labour/">https://www.business-humanrights.org/en/from-us/briefings/closing-the-gap-evidence-for-effective-human-rights-due-diligence-from-five-years-measuring-company-efforts-to-address-forced-labour/</a> (accessed 20 July 2022).
- Human Rights Rescource Center (n.d.), "List of large businesses, associations & investors with public statements & endorsements in support of mandatory due diligence regulation", available at: <a href="https://www.business-humanrights.org/en/latest-news/list-of-large-businesses-associations-investors-with-public-statements-endorsements-in-support-of-mandatory-due-diligence-regulation/">https://www.business-humanrights.org/en/latest-news/list-of-large-businesses-associations-investors-with-public-statements-endorsements-in-support-of-mandatory-due-diligence-regulation/</a> (accessed 20 July 2022).
- Human Rights Resource Center (2021), "Japan: Government reveals result of survey on Japanese companies' human rights initiatives in supply chains", available at <a href="https://www.business-humanrights.org/es/%C3%BAltimas-noticias/japan-govt-reveals-result-of-survey-on-japanese-companies-human-rights-initiatives-in-supply-chains/">https://www.business-human-rights-initiatives-in-supply-chains/</a> (accessed 20 July 2022).
- Kolstad I. (2016), "Too big to fault? Effects of the 2010 Nobel Peace Prize on Norwegian exports to China and foreign policy", CMI Working Paper WP No.3, Chr. Michelsen Institute, Bergen.
- Inter-American Commission of Human Rights (2016), "Indigenous Peoples, Afro-Descendent Communities and Natural Resources: Human Rights Protection in the Context of Extraction, Exploitation, and Development Activities". available at: <a href="https://www.oas.org/en/iachr/media\_center/preleases/2016/048.asp">https://www.oas.org/en/iachr/media\_center/preleases/2016/048.asp</a> (accessed 20 July 2022).
- Inter-American Commission of Human Rights (2019), "Business and Human Rights: Inter-American Standards", *Special Rapporteurship on Economic, Social, Cultural and Environmental Rights* REDESCAOEA/Ser.L/V/II CIDH/REDESCA/INF.1/19
  November 1st 2019
- Inter-American Court of Human Rights(2021), *Miskito Divers (Lemoth Morris et al) v Honduras*, available at <a href="https://media.business">https://media.business</a> humanrights.org/media/documents/seriec\_432\_ing.pdf (accessed 20 July 2022).
- IBIS Denmark (2013), "Guidelines for implementing indigenous peoples' right to free prior and informed consent", available at: http://www.socialimpactassessment.com/documents/Guidelines\_Implementing\_rights\_Indig enous Peoples FPIC.pdf (accessed 20 July 2022).
- Iglesias Márquez, D. (2020), "Estándares interamericanos sobre empresas y derechos humanos: Nuevas perspectivas para la conducta empresarial responsable en las Américas' *Anuario de Derechos Humanos* 347, Vol. 16, No. 2..
- International Centre for Human Rights and Democratic Development (n.d.), "Getting it right. Human rights impact assessment guide", available at: http://hria.equalit.ie/en/ (accessed 20 July 2022).
- International Corporate Accountability Roundtable, Danish Institute for Human Rights, and Global Business Initiative on Human Rights (2014), "Business dialogue on national action plans:

- report of key themes", available at: http://accountabilityroundtable.org/wp-content/uploads/2014/05/Business-Dialogue-ICAR-DIHR-GBI-Key-Themes.pdf (accessed 20 July 2022).
- International Council on Mining and Metals (2012), "Integrating human rights due diligence into corporate risk management processes", available at: http://www.icmm.com/page/75929/integrating-human-rights-due-diligence-into-corporate-risk-management-processes, (accessed 20 July 2022).
- International Federation for Human Rights (2011), "Community-based human rights impact assessments", available at: http://www.fidh.org/en/globalisation-human-rights/business-and-human-rights/7502-community-based-human-rights-impact-assessments (accessed 20 July 2022).
- International Business Leaders' Forum and International Finance Corporation (2010), *Guide to Human Rights Impact Assessment and Management*, available at: <a href="http://www.ifc.org/wps/wcm/connect/8ecd35004c0cb230884bc9ec6f601fe4/hriam-guide-092011.pdf?MOD=AJPERES">http://www.ifc.org/wps/wcm/connect/8ecd35004c0cb230884bc9ec6f601fe4/hriam-guide-092011.pdf?MOD=AJPERES</a> (accessed 20 July 2022).
- International Labour Organization (1989), Convention concerning indigenous and tribal peoples in independent countries (no. 169), available at: http://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPUB:12100:0::NO::P12100\_ILO\_COD E:C169 (accessed 20 July 2022).
- International Labour Organisation (1998), *Declaration on Fundamental Principles and Rights at Work*, available at: http://www.ilo.org/declaration/lang--en/index.htm (accessed 20 July 2022).
- International Labour Organisation (2006), *Tripartite Declaration of Principles concerning Multinational Enterprises and Social Policy*, available at: http://www.ilo.org/wcmsp5/groups/public/---ed\_emp/---emp\_ent/---multi/documents/publication/wcms\_094386.pdf (accessed 20 July 2022).
- International Labour Organization (2013), "The Rana Plaza Building Collapse...100 days on", available at: <a href="http://www.ilo.org/global/about-the-ilo/activities/all/WCMS\_218693/lang-en/index.htm">http://www.ilo.org/global/about-the-ilo/activities/all/WCMS\_218693/lang-en/index.htm</a> (accessed 20 July 2022).
- International Labour Organization (2017), "Tripartite Declaration of Principles Concerning Multinational Enterprises and Social Policy" para 10 (d).
- International Standards Organisation (2010), "International Standard 26000, *Guidance on Social Responsibility*" (Geneva: ISO).
- International Labour Organisation (2020), "Achieving decent work in global supply chainsTMDWSC/2020 Report for discussion at the technical meeting on achieving decent work in global supply chains", available at: https://www.ilo.org/wcmsp5/groups/public/---ed\_dialogue/---dialogue/documents/meetingdocument/wcms\_736541.pdf

  https://www.ilo.org/wcmsp5/groups/public/---ed\_dialogue/--dialogue/documents/meetingdocument/wcms\_736541.pdf (accessed 20 July 2022).
- IPIECA and Danish Institute for Human Rights (2013), "Integrating human rights into environmental, social and health impact assessments. A practical guide for the oil and gas industry", available at: http://www.humanrights.dk/files/media/dokumenter/tools/integrating\_hr\_into\_eshia.pdf (accessed 20 July 2022).
- Islam, M. and McPhail, K. (2011), "Regulating for corporate human rights abuses: the emergence of corporate reporting on the ILO's human rights standards within the global garment manufacturing and retail industry", *Critical Perspectives on Accounting*, Vol. 22 No. 8, pp. 790-810.

- Kaspersen, M. and Johansen, T.R. (2014), "Changing social and environmental reporting systems", *Journal of Business Ethics*, 13 December, pp. 1-19.
- Kenner, J., & Peake, K. (2017). "The Bangladesh Sustainability Compact: An Effective Exercise of Global Experimentalist EU Governance?", *Cambridge Yearbook of European Legal Studies*, No.19, pp. 86-115.
- Klonick K., (2019), "The Facebook Oversight Board: Creating an Independent Institution to Adjudicate Online Free Expression", Yale Law JournalNo. 129, 2418 (2019-2020), available at:

  <a href="https://heinonline.org/HOL/LandingPage?handle=hein.journals/ylr129&div=48&id=&page=(accessed 20 July 2022).">https://heinonline.org/HOL/LandingPage?handle=hein.journals/ylr129&div=48&id=&page=(accessed 20 July 2022).</a>
- Knapp V.(2022), "Proposed EU Directive on Corporate Sustainability Due Diligence: Why Non-EU Businesses Should Respond to the Consultation", *Oxford Business Law Blog*, available at: <a href="https://www.law.ox.ac.uk/business-law-blog/blog/2022/05/proposed-eu-directive-corporate-sustainability-due-diligence-why-non">https://www.law.ox.ac.uk/business-law-blog/blog/2022/05/proposed-eu-directive-corporate-sustainability-due-diligence-why-non">https://www.law.ox.ac.uk/business-law-blog/blog/2022/05/proposed-eu-directive-corporate-sustainability-due-diligence-why-non">https://www.law.ox.ac.uk/business-law-blog/blog/2022/05/proposed-eu-directive-corporate-sustainability-due-diligence-why-non">https://www.law.ox.ac.uk/business-law-blog/blog/2022/05/proposed-eu-directive-corporate-sustainability-due-diligence-why-non">https://www.law.ox.ac.uk/business-law-blog/blog/2022/05/proposed-eu-directive-corporate-sustainability-due-diligence-why-non">https://www.law.ox.ac.uk/business-law-blog/blog/2022/05/proposed-eu-directive-corporate-sustainability-due-diligence-why-non">https://www.law.ox.ac.uk/business-law-blog/blog/2022/05/proposed-eu-directive-corporate-sustainability-due-diligence-why-non</a> (accessed 20 July 2022).
- Lebaron, G., & Lister, J. (2015), "Benchmarking global supply chains: The power of the 'ethical audit' regime" *Review of International Studies*, Vol.41 No.5, pp. 905-924. doi:10.1017/S0260210515000388
- Lebaron, G. (2021) "Wages: An Overlooked Dimension of Business and Human Rights in Global Supply Chains", *Business and Human Rights Journal*, Vol. 6 No.1, pp. 1-20. doi:10.1017/bhj.2020.32
- LeBaron G. & Lister, J. (2022), "The hidden costs of global supply chain solutions", *Review of International Political Economy*, Vol. 29 No.3, pp. 669-695, doi: 10.1080/09692290.2021.1956993
- Lim A. (2017), "Global corporate responsibility disclosure: A comparative analysis of field, national, and global influences" *International Sociology*. Vol. 32 No. 1, pp-61-85. doi:10.1177/0268580916673748
- Lim, A. & Tsutsui, K. (2012), "Globalization and Commitment in Corporate Social Responsibility: Cross-National Analyses of Institutional and Political-Economy Effects" *American Sociological Review* No. 77, pp.69-98. doi 10.1177/0003122411432701.
- <u>Littenberg, Elliot, Witschi, (2022),</u> "Complying with the Uyghur Forced Labor Prevention Act a Detailed Compliance Roadmap", *Ropes & Grey*, available at: <a href="https://www.ropesgray.com/en/newsroom/alerts/2022/06/Complying-with-the-Uyghur-Forced-Labor-Prevention-Act-a-Detailed-Compliance-Roadmap">https://www.ropesgray.com/en/newsroom/alerts/2022/06/Complying-with-the-Uyghur-Forced-Labor-Prevention-Act-a-Detailed-Compliance-Roadmap (accessed 19 July 2022).
- Londono-Lozano, M. et al (2017), The Inter-American court of human rights and multinational enterprises: towards business and human rights in the Americas? The Law & Practice of International Courts and Tribunals available at <a href="https://brill.com/view/journals/lape/16/3/article-p437\_437.xml">https://brill.com/view/journals/lape/16/3/article-p437\_437.xml</a> (accessed 20 July 2022).
- Maher, R. (2020)"De-contextualized Corporate Human Rights Benchmarks: Whose Perspective Counts? See Disclaimer", *Business and Human Rights Journal*, Vol.5 No.1, pp.156-163. doi:10.1017/bhj.2019.19
- Mantouvalou, V. (2018), "The UK Modern Slavery Act 2015: Three Years On". *The Modern Law Review*, No. 81, pp,1017-1045. https://doi.org/10.1111/1468-2230.12377
- Marotta, F. (2013), "Request from the Chair of the OECD Working Party on responsible business conduct", *United Nations Human Rights Office of the High Commissioner*, 27 November, available at: http://www.ohchr.org/Documents/Issues/Business/LetterOECD.pdf (accessed 20 July 2022).
- Marquis, C., Toffel, M.W., and Zhou, Y. (2011), "Scrutiny, norms, and selective disclosure: a global study of greenwashing", Working Paper No. 11-115, Harvard Business School

- Organizational Behavior Unit, available at: https://ideas.repec.org/p/hbs/wpaper/11-115.html (accessed 20 July 2022).
- Martin-Ortega, Olga. (2013). "Human Rights Due Diligence for Corporations: From Voluntary Standards to Hard Law at Last?". *Netherlands Quarterly of Human Rights, Vol.*32, pp. 44-74. Doi 10.1177/016934411403200104.
- Martin-Ortega, O. (2017), "Human Rights Risks in Global Supply Chains: Applying the UK Modern Slavery Act to the Public Sector". *Glob Policy*, Vol.8, pp. 512-521. https://doi.org/10.1111/1758-5899.12501
- Martin-Ortega, O., (2018), "Public Procurement as a Tool for the Protection and Promotion of Human Rights: a Study of Collaboration, Due Diligence and Leverage in the Electronics Industry", *Business and Human Rights Journal*, Cambridge University Press, vol. 3 No.1, pp 75-95.
- McPhail, K.J. and Adams, C.A. (2016), "Corporate Respect For Human Rights: Meaning, Scope, And The Shifting Order Of Discourse", *Accounting, Auditing And Accountability Journal.*, Vol. 29, No.4, pp. 650-678.
- McCorquodale, R. (2022), "Human Rights Due Diligence Instruments: Evaluating The Current Legislative
  - Landscape" in Marx, A. (Ed.), 2022, Research Handbook on Global Governance, Business and Human Rights, Edward Elgar Publishing, Cheltenham.
- McCorquodale, R., Nolan, J. (2021), "The Effectiveness of Human Rights Due Diligence for Preventing Business Human Rights Abuses", *Neth Int Law Rev*, Vol.68, pp. 455–478 available at: <a href="https://doi.org/10.1007/s40802-021-00201-x">https://doi.org/10.1007/s40802-021-00201-x</a> (accessed 20 July 2022).
- McCorquodale, R., Smit L, Neely S, (2017), "Human Rights Due Diligence in Law and Practice: Good Practices and Challenges for Business Enterprises, *Business and Human Rights Journal*, Vol. 2, pp.195–224
- Marslev, K. (2019), "The Political Economy of Social Upgrading: A class-relational analysis of social and economic trajectories of the garment industries of Cambodia and Vietnam", Roskilde: Roskilde Universitet. p. 271
- McVey M., Ferguson J., PuyouF.R., 2022, "Traduttore, Traditore?" Translating Human Rights into the Corporate Context", *Journal of Business Ethics*, available at: https://doi.org/10.1007/s10551-021-05028-3 (accessed 20 July 2022).
- Methven O'Brien, C. (2009), *Human rights and Transnational Corporations: For a multi-level governance approach*. Ph.D. thesis. European University Institute, Florence, available at <a href="http://hdl.handle.net/1814/12986">http://hdl.handle.net/1814/12986</a> (accessed 20 July 2022).
- Methven O'Brien, C. (2011), "The UN Special Representative on Business and Human Rights: Reembedding or Dis-embedding Transnational Markets", Ch.13 in Joerges C. and Falke, J., Polanyi, K.. (Eds.), 2011, "Globalisation and the Potential of Law in Transnational Markets" Oxford Hart.
- Methven O'Brien, C. (2016), "Guest Post: Claire Methven O'Brien on Ireland's Working Outline for a National Plan on Business and Human Rights", *Business and Human Rights in Ireland Blog*, 28 January 2016, available at: https://businesshumanrightsireland.wordpress.com/2016/01/28/guest-post-claire-methven-obrien-on-irelands-working-outline-for-a-national-plan-on-business-and-human-rights/ (accessed 18 July 2022)."
- Methven O'Brien, C. (2019), *Business and Human Rights. A Handbook for Legal Practitioners*, Council of Europe, Strasbourg.

- Methven O'Brien, C., (2020a) "Transcending the Binary: Linking Hard and Soft Law Through a UNGPS-Based Framework Convention", *American Journal of International Law Unbound*, No. 114, pp.186-191.
- Methven O'Brien C., (2020 b), "Confronting the constraints of the medium: The Fifth Session of the UN Intergovernmental Working Group on a Business and Human Rights Treaty", 5(1) Business and Human Rights Journal Vol. 5, No, 1, pp.150-155.
- Methven O' Brien C., (2021), "Business and human rights in Europe 2011-2021: A decade in review", in Philip Czech et al (eds.), 2021, *European Yearbook of Human Rights*, Intersentia, PP. 407-437.
- Methven O'Brien (2022), "Bounded Rationality, Metonymy, Humility: Further Arguments for a Framework-Style Business and Human Rights Treaty", Völkerrechtsblog, doi: 10.17176/20220622-153107-0.
- Methven O'Brien, C. and Dhanarajan, S. (2016), "The corporate responsibility to respect human rights: a status review", *Accounting, Auditing & Accountability Journal*, Vol. 29 No. 4, pp. 542-567. https://doi.org/10.1108/AAAJ-09-2015-2230
- Methven O'Brien, C., Mehra, A., Blackwell, S., and Poulsen Hansen, C.B., (2014), "National Action Plans on Business and Human Rights. A Toolkit for the Development, Implementation and Review of State Commitments to Business and Human Rights Frameworks", DIHR/ICAR, Copenhagen/Washington DC, available at:
- http://www.humanrights.dk/publications/national-action-plans-business-human-rights (accessed 20 July 2022).
- Methven O'Brien, C., Mehra, A., Blackwell, S., & Poulsen-Hansen, C. (2016). "National Action Plans: Current Status and Future Prospects for a New Business and Human Rights Governance Tool". *Business and Human Rights Journal*, *1*(1), 117-126. doi:10.1017/bhj.2015.14, available at: http://journals.cambridge.org/action/displayAbstract?fromPage=online&aid=10006475&full textType=CR&fileId=S2057019815000140 (accessed 20 July 2022).
- Methven O' Brien C., Martin-Ortega O., (2018) Discretion, Divergence, Paradox: Public and private Supply Chain Standards on Human Rights, Social Science Research Network Available at SSRN: <a href="https://ssrn.com/abstract=3197615">https://ssrn.com/abstract=3197615</a>
- Methven O'Brien, Martin-Ortega (2020), "EU Human Rights Due Diligence Legislation: Monitoring, Enforcement and Access to Justice for Victims. Options for the EU: Briefing 2, June 2020", Policy Department for External Relations, Directorate General for External Policies of the European Parliament,603.505, available at: <a href="https://www.europarl.europa.eu/thinktank/en/document/EXPO\_BRI%282020%29603">https://www.europarl.europa.eu/thinktank/en/document/EXPO\_BRI%282020%29603</a> 505 (accessed 20 July 2022).
- Methven O'Brien, C, and Martin-Ortega, O. (2020), "Human rights and public procurement of goods and services", in Deva S. and Birchall D. (eds), 2020, Research Handbook on Human Rights and Business, , pp. 245–267 available at https://doi.org/10.4337/9781786436405.00019
- Methven O' Brien C., Martin-Ortega O., (2022), "Commission Proposal on Corporate Sustainability Due Diligence: Analysis From A Human Rights Perspective", Policy Department for External Relations Directorate General for External Policies of the Union.
- Ministry of Foreign Affairs, the Ministry of Economy, Trade and Industry (METI), Government of Japan, (2021), "Release on the Results from the Questionnaire Survey on the Status of Efforts on Human Rights in the Supply Chains of Japanese Companies", available at <a href="https://www.meti.go.jp/english/press/2021/1130\_002.html">https://www.meti.go.jp/english/press/2021/1130\_002.html</a> (accessed 19 July 2022)
- Moyn, S. (2012). The Last Utopia: Human Rights in History. Cambridge, Mass.

- National Human Rights Council of Brazil (2020), "National guidelines for the adoption of a Business and Human Rights policy", available at http://homacdhe.com/wpcontent/uploads/2020/03/Resolu%C3%A7%C3%A3o-n%C2%BA5-2020-CNDH.pdf. (accessed 20 July 2022).
- Nestlé Global (n.d.), "Human rights", available at: http://www.nestle.com/csv/human-rights-compliance/human-rights (accessed 18 July 2022).
- Neumayer E. (2005), "Do International Human Rights Treaties Improve Respect for Human Rights?", *Journal of Conflict Resolution*, Vol. 49 No.6, pp.925–953. https://doi.org/10.1177/0022002705281667.
- New Zealand Ministry of Business, Innovation and Employment (2022), "Consultation on Modern Slavery and Worker Exploitation", available at <a href="https://www.mbie.govt.nz/have-your-say/modern-slavery/">https://www.mbie.govt.nz/have-your-say/modern-slavery/</a> (accessed 20 July 2022)
- Nolan, J., Frishling N., (2020) Human Rights Due Diligence and the (Over)-Reliance on Social Auditing in Supply Chains, Justine Nolan and Nana Frishling in Deva S., Brichall D., Research Handbook on Human Rights and Business
- Nolan J (2022), "Chasing the next shiny thing: Can human rights due diligence effectively address labour exploitation in global fashion supply chains?", *International Journal for Crime*, *Justice*
- and Social Democracy. Vol 11No.2pp. 1-14, available at: <a href="https://doi.org/10.5204/ijcjsd.2398">https://doi.org/10.5204/ijcjsd.2398</a> NomoGaia (n.d.), "Human rights impact assessment toolkit", available at: <a href="http://nomogaia.org/tools/">http://nomogaia.org/tools/</a> (accessed 20 July 2022).
- Narine M (2015), "Disclosing disclosure's defects: addressing corporate responsibility for human rights impacts" *Columbia Law Rev* Vol. 47 No1, pp. 84–150.
- Novak Talavera, F., (2014), "Corporate Social Responsibility in the Area of Human Rights and Environment in the Americas", 84th Regular Session OEA/Ser.Q, available at <a href="http://www.oas.org/en/sla/iajc/docs/cji-doc\_449-14\_rev1\_en.pdf">http://www.oas.org/en/sla/iajc/docs/cji-doc\_449-14\_rev1\_en.pdf</a> (accessed 20 July 2022).
- O'Connor, C., Labowitz, S. (2017), "Putting the 'S' in ESG: measuring human rights performance for investors", NYU Stern Center for Business and Human Rights, New York, available at: <a href="https://static1.squarespace.com/static/547df270e4b0ba184dfc490e/t/58cad912e58c6274180b58b6/1489688854754/Metrics-Report-final-1.pdf?utm\_source=Segment+2++VIP+suppressed&utm\_campaign=bb69a89595EMAIL\_CAMPAIGN\_2017\_03\_16&utm\_medium=email&utm\_term=0\_59dce0b125bb69a89595-505931093 (accessed 20 July 2022)
- O'Rourke, D. (2002), "Monitoring the monitors: A critique of corporate third-party labor monitoring" *Corporate responsibility and labour rights: Codes of conduct in the Global Economy Journal*, Earthscan, London, available at <a href="https://scholar.google.com/citations?view\_op=view\_citation&hl=en&user=tJ3oJEkAAAAJ">https://scholar.google.com/citations?view\_op=view\_citation&hl=en&user=tJ3oJEkAAAAJ</a> & <a href="https://scholar.google.com/citations?view\_op=view\_citation&hl=en&user=tJ3oJEkAAAAJ</a> & <a href="https://scholar.google.com/citations">https://scholar.google.com/citations?view\_op=view\_citation&hl=en&user=tJ3oJEkAAAAJ</a> & <a href="https://scholar.google.com/citations">https://scholar.google.com/citations?view\_op=view\_citation&hl=en&user=tJ3oJEkAAAAJ</a> & <a href="https://scholar.google.com/citations">https://scholar.google.com/citations</a>? <a href="https://scholar.google.com/citations">https://scholar.google.com/citations</a>? <a href="https://scholar.google.com/citations">https://scholar.google.com/citations</a>? <a href="https://scholar.google.com/citations">https://scholar.google.com/citations</a> <a href="https://scholar.google.com/citations">https://scholar.google.com/citations</a> <a href="https://scholar.google.com/citations">https://scholar.google.com/citations</a> <a href="https://scholar.google.com/citations">https://scholar.google.com/citations</a> <a href="https://scholar.google.com/citations">https://scholar.google.com/citations</a> <a href="https://scholar.google.com/citations">https://scholar.google.com/citations</a> <a href="https://scholar.google.com/citations">https://scholar.google.com/citati
- Office of the High Commissioner for Human Rights, (2016), "Guidance on National Action Plans on Business and Human Rights", available at <a href="https://www.ohchr.org/sites/default/files/Documents/Issues/Business/UNWG\_NAPGuidance.pdf">https://www.ohchr.org/sites/default/files/Documents/Issues/Business/UNWG\_NAPGuidance.pdf</a> (accessed 20 July 2022).
- Office of the High Commissioner of Human Rights, (2016), "Accountability and Remedy Project", accessible at <a href="https://www.ohchr.org/EN/Issues/Business/Pages/OHCHRaccountabilityandremedyproject.aspx">https://www.ohchr.org/EN/Issues/Business/Pages/OHCHRaccountabilityandremedyproject.aspx</a> (accessed 20 July 2022).

- Office of the High Commissioner of Human Rights, (2018) "General Recommendation No. 37 on gender-related dimensions of disaster risk reduction in a changing climate", available at <a href="https://www.ohchr.org/en/documents/general-comments-and-recommendations/general-recommendation-no37-2018-gender-related">https://www.ohchr.org/en/documents/general-comments-and-recommendations/general-recommendation-no37-2018-gender-related</a> (accessed 20 July 2022).
- Office of the High Commissioner of Human Rights, (2021), "Human rights-compatible international investment agreements", Note by the Secretary-General, available at <a href="https://documents-dds-ny.un.org/doc/UNDOC/GEN/N21/208/09/PDF/N2120809.pdf">https://documents-dds-ny.un.org/doc/UNDOC/GEN/N21/208/09/PDF/N2120809.pdf</a>?OpenElement (accessed 20 July 2022).
- Office of the High Commissioner of Human Rights, (2022 a), "Panel discussion on the tenth anniversary of the Guiding Principles on Business and Human Rights", available at <a href="https://www.ohchr.org/sites/default/files/2022-02/report-panel-10th-anniversary-ungps.pdf">https://www.ohchr.org/sites/default/files/2022-02/report-panel-10th-anniversary-ungps.pdf</a> (accessed 20 July 2022).
- Office of the High Commissioner of Human Rights, (2022 b), "Sustainable Global Supply Chains: G7 Leadership on UNGP Implementation", available at <a href="https://www.ohchr.org/sites/default/files/2022-05/report-sustainable-global-supply-chains-g7.pdf">https://www.ohchr.org/sites/default/files/2022-05/report-sustainable-global-supply-chains-g7.pdf</a> (accessed 20 July 2022).
- Okowa, P. (2020), "The Pitfalls Of Unilateral Legislation In International Law: Lessons From Conflict Minerals Legislation" *International and Comparative Law Quarterly*, Vol 69 No.3, pp. 685-717. doi:10.1017/S0020589320000238
- Organisation for Economic Development and Co-operation, 2018, 'Due Diligence Guidance for Responsible Business Conduct', available at <a href="https://mneguidelines.oecd.org/due-diligence-guidance-for-responsible-business-conduct.htm">https://mneguidelines.oecd.org/due-diligence-guidance-for-responsible-business-conduct.htm</a>, (accessed 20 July 2022).
- Organisation for Economic Development and Co-operation (2011), "OECD Guidelines for Multinational Enterprises", available at: http://mneguidelines.oecd.org/text/ (accessed 20 July 2022).
- Organisation for Economic Development and Co-operation (2013), "Due diligence guidance for responsible supply chains of minerals from conflict-affected and high-risk areas", available at: http://www.oecd-ilibrary.org/content/book/9789264185050-en (accessed 20 July 2022).
- Organisation for Economic Co-operation and Development (2016), "Due Diligence Guidance for Responsible Supply Chains of Minerals from Conflict-Affected and High-Risk Areas", available at: <a href="https://www.oecd.org/corporate/mne/mining.htm">https://www.oecd.org/corporate/mne/mining.htm</a>, (accessed 20 July 2022).
- Organisation for Economic Development and Co-operation (2018) "Due Diligence Guidance for Responsible Business Conduct", available at: <a href="https://www.oecd.org/investment/due-diligence-guidance-for-responsible-business-conduct.htm">https://www.oecd.org/investment/due-diligence-guidance-for-responsible-business-conduct.htm</a> (accessed 20 July 2022).
- Organisation for Economic Development and Co-operation (2020), "Responsible Business Conduct Country Fact Sheet BRAZIL" available at: <a href="https://mneguidelines.oecd.org/responsible-business-conduct-country-fact-sheet-brazil.pdf">https://mneguidelines.oecd.org/responsible-business-conduct-country-fact-sheet-brazil.pdf</a> (accessed 20 July 2022).
- Organisation for Economic Development and Co-operation Watch (2014), "CSRD et al vs Andritz AG", available at: http://www.oecdwatch.org/cases/Case 326 (accessed 20 July 2022 ).
- Parliament of Canada, *Bill S-216 An Act to enact the Modern Slavery Act and to amend the Customs Tariff* available at: <a href="https://parl.ca/DocumentViewer/en/43-2/bill/S-216/first-reading">https://parl.ca/DocumentViewer/en/43-2/bill/S-216/first-reading</a> (accessed 20 July 2022).
- Park, (2018a), "Investors as Regulators: Green Bonds and the Governance Challenges of the Sustainable Finance Revolution", Stanford Journal of International Law Vol. 54 No1
- Park, S. (2018b), "Social Bonds for Sustainable Development: A Human Rights Perspective on Impact Investing". *Business and Human Rights Journal*, Vol 3 No2, pp. 233-255. doi:10.1017/bhj.2018.6 (accessed 20 July 2022).

- Pietropaoli, I., Johnstone, O., Balch A., (2021), "Effectiveness Of Forced Labour Import Bans Policy Brief, Modern Slavery PEC Policy Brief 2021", available at: <a href="https://modernslaverypec.org/assets/downloads/PEC-Policy-Brief-Effectiveness-Forced-Labour-Import-Bans.pdf">https://modernslaverypec.org/assets/downloads/PEC-Policy-Brief-Effectiveness-Forced-Labour-Import-Bans.pdf</a>, (accessed 20 July 2022).
- Pope S., Lim A. (2020), "Three types of organizational boundary spanning: Predicting CSR policy extensiveness among global consumer products companies" *Business Ethics: A European Review*, Vol. 29 No.3, pp. 451–470.
- Pope, S, & Lim, A. (2022), "Why companies practice corporate social responsibility" *Mit Sloan Management Review*, Vol.1No.3available at: <a href="https://ink.library.smu.edu.sg/soss\_research/3523">https://ink.library.smu.edu.sg/soss\_research/3523</a> (accessed 20 July 2022).
- Power, M. (2004). The Risk Management of Everything. Journal of Risk Finance, The. 5. 58-65. 10.1108/eb023001. available athttps://www.researchgate.net/publication/242344966\_The\_Risk\_Management\_of\_Everything.
- Posner, E. (2014), *The Twilight of Human Rights Law*, Oxford University Press, available at <a href="https://link.springer.com/article/10.1007/s12142-021-00637-x#ref-CR45">https://link.springer.com/article/10.1007/s12142-021-00637-x#ref-CR45</a> (accessed 20 July 2022).
- Presidência da República, (2018) "Decreto Nº 9.571, De 21 De Novembro De 2018, National Guidelines on Business and Human Rights", available at <a href="http://www.planalto.gov.br/ccivil\_03/\_ato2015-2018/2018/decreto/D9571.htm">http://www.planalto.gov.br/ccivil\_03/\_ato2015-2018/2018/decreto/D9571.htm</a> (accessed 20 July 2022)
- Preuss, L. and Brown, D. (2012), "Business Policies on Human Rights: An Analysis of Their Content and Prevalence Among FTSE 100 Firms", *Journal of Business Ethics* Vol.109, Issue 3, pp.289-299.
- Rahim, M. (2020), Humanizing the Global Supply Chain: Building a 'Decent Work' Environment in the Ready-made Garments Supply Industry in Bangladesh in Deva S., Brichall D.,(eds,) 2020, Research Handbook on Human Rights and Business, Elgar Edward Publishing.
- République Française (2012), *Décret n° 2012-557 du 24 avril 2012 relatif aux obligations de transparence des entreprises en matière sociale et environnementale*, available at: <a href="https://www.legifrance.gouv.fr/affichTexte.do?cidTexte=JORFTEXT000025746900&categorieLien=id">https://www.legifrance.gouv.fr/affichTexte.do?cidTexte=JORFTEXT000025746900&categorieLien=id</a> (accessed 20 July 2022).
- République Française (2017), LOI n° 2017-399 du 27 mars 2017 relative au devoir de vigilance des sociétés mères et des entreprises donneuses d'ordre, available at <a href="https://www.legifrance.gouv.fr/loda/id/JORFTEXT000034290626/">https://www.legifrance.gouv.fr/loda/id/JORFTEXT000034290626/</a> (accessed 20 July 2022).
- Rivera, C. (2018), "Business and Human Rights in the Americas: Defining a Latin American Route to Corporate Responsibility" in Letnar Cernic, J. and Carrillo Santarelli N. (Eds.), 2018, *The Future of Business and Human Rights Theoretical and Practical Considerations for a UN Treaty*, Intersentia, pp. 163-184 https://doi.org/10.1017/9781780686455.009
- Ruggie, J.G. (2008), "Response Letter to Arvind Ganesan of Human Rights Watch" (28 Jan 2008), available at: http://business-humanrights.org/en/ngos-criticise-un-special-representative-ruggies-draft-guiding-principles-on-business-human-rights#c56768 (accessed 20 July 2022).
- Ruggie, J.G. (2013), *Just business: multinational corporations and human rights*. First edition. New York: W. W. Norton & Company.
- Ruggie, J.G. and Nelson, T. (2015), "Human Rights and the OECD Guidelines for Multinational Enterprises: Normative Innovations and Implementation Challenges", Corporate Social Responsibility Initiative Working Paper No.66, available at:

- http://shiftproject.org/sites/default/files/Ruggie-Nelson\_OECDNCPanalysis\_May2015.pdf (accessed 20 July 2022).
- Ruggie, J.G., Sherman, J., (2017), "The Concept of 'Due Diligence' in the UN Guiding Principles on Business and Human Rights: A Reply to Jonathan Bonnitcha and Robert McCorquodale", European Journal of International Law, Vol. 28, No.3 pp. 921–928, <a href="https://doi.org/10.1093/ejil/chx047">https://doi.org/10.1093/ejil/chx047</a>.
- Ruggie J.G., (2020), "The Paradox of Corporate Globalization: Disembedding and Reembedding Governing Norms", M-RCBG Faculty Working Paper Series, Mossavar-Rahmani Center for Business & Government, Harvard Kennedy School, available at <a href="https://scholar.harvard.edu/files/john-ruggie/files/the\_paradox\_of\_corporate\_globalization.pdf">https://scholar.harvard.edu/files/john-ruggie/files/the\_paradox\_of\_corporate\_globalization.pdf</a> (accessed 20 July 2022)
- Russell, K., Gates, G., Keller, J. and Watkins, D. (2016) "How Volkswagen Got Away with Diesel Deception" *New York Times*, available at http://www.nytimes.com/interactive/2015/business/international/vw-diesel-emissions-scandal-explained.html? r=0 (accessed 20 July 2022).
- Salcito, K., (2019), "Company-commissioned HRIA: Concepts, practice, limitations and Opportunities" in Götzmann (Ed.) 2019, *Handbook on Human Rights Impact Assessment*available at https://doi.org/10.4337/9781788970006.00010
- Sanabria, D.; Schönfelder, D. (2021), "Recognising Nuances: Mandatory Human Rights Due Diligence in Mexico and Colombia", VerfBlog, available at: https://verfassungsblog.de/recognising-nuances/, DOI: 10.17176/20210421-101319-0. (accessed 20 July 2022).
- Save the Children, (2016), "Children's rights: the Ultimate definition of sustainability", Stockholm: Save the Children
- Schweizerische Eidgnenossenschaft, (2021), Verordnung über Sorgfaltspflichten und Transparenz bezüglich Mineralien und Metallen aus Konfliktgebieten und Kinderarbeit, available at https://www.fedlex.admin.ch/eli/cc/2021/847/de (accessed 20 July 2022)
- Schaper S., Pollach I., (2021) "Modern slavery statements: From regulation to substantive supply chain reporting", *Journal of Cleaner Production*, Vol.No.313, 127872, available at:https://www.sciencedirect.com/science/article/pii/S0959652621020904 Schilling-Vacaflor, A. (2012), "Democratizing resource governance through prior consultations? Lessons from Bolivia's hydrocarbon sector", Working Paper No. 184, German Institute for Global and Area Studies, Hamburg,, available at: http://www.giga-hamburg.de/en/system/files/publications/wp184\_schilling.pdf (accessed 20 July 2022)
- SEBI, (2021), "Extension of applicability of Business Responsibility Reporting (BRRs) to top 1000 listed entities from present requirement to 500 listed entities", available at <a href="https://www.sebi.gov.in/sebi\_data/meetingfiles/dec-2019/1576469077048\_1.pdf">https://www.sebi.gov.in/sebi\_data/meetingfiles/dec-2019/1576469077048\_1.pdf</a> (accessed 20 July 2022)
- Sedex (2014), "Sedex Global partners with World Bank Institute to develop open supply chain platform", available at: http://www.sedexglobal.com/world-bank-institute-partners-with-sedex-global-to-develop-open-supply-chain-platform/ (accessed 20 July 2022).
- Serva C., Faria, L. (2022), "Mandatory human rights due diligence in Brazil", *International Bar Association*, available at <a href="https://www.ibanet.org/Mandatory-human-rights-due-diligence-Brazil#\_edn1">https://www.ibanet.org/Mandatory-human-rights-due-diligence-Brazil#\_edn1</a> (accessed 20 July 2022).
- Schliemann, C., Bryk L., (2019), "Arms Trade And Corporat Responsibility

- Liability, Litigation and Legislative Reform", *Friedrich-Ebert-Stiftung* Study, available at https://library.fes.de/pdf-files/iez/15850.pdf (accessed 20 July 2022)
- Schilling-Vacaflor, A. (2020) "Putting the French Duty of Vigilance Law in Context: Towards Corporate Accountability for Human Rights Violations in the Global South?", *Human Rights Review*, Vol.23. No.1, pp.41-41.
- Sherpa et al, (2019), "The Law on Duty of Vigilance of Parent and Outsourcing Companies, Year 1:
  - Companies Must Do Better", available at <a href="https://www.asso-sherpa.org/wpcontent/uploads/2019/06/2019.06.14-EN-Rapport-Commun-Companies-must-do-better\_compressed\_compressed-1.pdf">https://www.asso-sherpa.org/wpcontent/uploads/2019/06/2019.06.14-EN-Rapport-Commun-Companies-must-do-better\_compressed\_compressed-1.pdf</a> (accessed 20 July 2022).
- Sherpa, Terre Solidaire (2021) "RADAR Duty of Vigilance Radar Report 2021", available at <a href="https://plan-vigilance.org/wp-content/uploads/2021/07/2021-07-05-Radar-DDV-Rapport-2021.pdf">https://plan-vigilance.org/wp-content/uploads/2021/07/2021-07-05-Radar-DDV-Rapport-2021.pdf</a> (accessed 19 July 2022)
- Shift (2013a), "From audit to innovation: advancing human rights in global supply chains", available at: http://shiftproject.org/sites/default/files/From%20Audit%20to%20Innovation-Advancing%20Human%20Rights%20in%20Global%20Supply%20Chains\_0.pdf (accessed 20 July 2022).
- Shift (2013b), "Using leverage in business relationships to reduce human rights risks", available at: http://shiftproject.org/sites/default/files/Using%20Leverage%20in%20Business%20Relation ships%20to%20Reduce%20Human%20Rights%20Risks.pdf (accessed 20 July 2022).
- Shift (2019), "Human rights reporting in France: Two Years In: Has the Duty of Vigilance Law led to more Meaningful Disclosure?" available at <a href="https://shiftproject.org/wp-content/uploads/2019/11/Shift\_HumanRightsReportinginFrance\_Nov27.pdf">https://shiftproject.org/wp-content/uploads/2019/11/Shift\_HumanRightsReportinginFrance\_Nov27.pdf</a> (accessed 20 July 2022).
- Siddiqui, J. & Uddin, S. (2016) "Human Rights Disasters, Corporate Accountability And The State: An Emerging Economy Case" *Accounting Auditing & Accountability Journal*. Vol.29, pp- 679 704, doi 10.1108/AAAJ-07-2015-2140.
- Smit, L. Holly, McCorquodale R. and Neely S.,(2021), "Human rights due diligence in global
- supply chains: evidence of corporate practices to inform a legal standard", *International Journal of Human Rights*, Vol.25No.6, pp. 945-973, DOI: 10.1080/13642987.2020.1799196.
- Smit, L., Bright, C., Pietropaoli, I., Hughes-Jennett, J., & Hood, P. (2020) "Business Views on Mandatory Human Rights Due Diligence Regulation: A Comparative Analysis of Two Recent Studies" *Business and Human Rights Journal*, Vol. 5 No.2, pp.261-269. doi:10.1017/bhj.2020.10.
- State of California, Department of Justice, (2012), *California Transparency in Supply Chains Act*, available at <a href="https://oag.ca.gov/SB657">https://oag.ca.gov/SB657</a> (accessed 15 July 2022).
- Steinweg, T. and Ten Kate, G. (2013), "Sourcing of minerals could link EU companies to violent conflict", *Centre for Research on Multinational Corporations*, available at: http://www.somo.nl/news-en/sourcing-of-minerals-could-link-eu-companies-to-violent-conflict (accessed 20 July 2022).
- Stevenson, M. and Cole, R. (2018), "Modern slavery in supply chains: a secondary data analysis of detection, remediation and disclosure", *Supply Chain Management*, Vol. 12 No. 3, pp. 81-99. https://doi.org/10.1108/SCM-11-2017-0382.

- Stoop N, Verpoorten M, Van Der Windt P. (2018) "More legislation, more violence? The impact of Dodd-Frank in the DRC". *PLoS ONE* Vol.13 No.8available at <a href="https://doi.org/10.1371/journal.pone.0201783">https://doi.org/10.1371/journal.pone.0201783</a> (accessed 20 July 2022).
- Storvinget, 2021, *Vedtak til lov om virksomheters åpenhet og arbeid med grunnleggende menneskerettigheter og anstendige arbeidsforhold (åpenhetsloven)*, available at https://www.stortinget.no/no/Saker-og-publikasjoner/Vedtak/Beslutninger/Lovvedtak/2020-2021/vedtak-202021-176/ (accessed 20 July 2022).
- Subasinghe, R. (2021) "A Neatly Engineered Stalemate: A Review of the Sixth Session of Negotiations on a Treaty on Business and Human Rights", *Business and Human Rights Journal*, Vol.6 No.2, pp.384-391. doi:10.1017/bhj.2021.17
- SUTREL Sindicato Unificado de Trabajadores de la Electricidad y Actividades Conexas de Lima y Callao, (n.d.), available at: <a href="https://publicservices.international/affiliate/sutrel/videos?search=%7B%22types%22%3A%5B%22video%22%5D,%22tags%22%3A%7B%22type%22%3A%5B%5D,%22ally%22%3A%5B%5D%7D%7D&lang=en&affiliate\_id=708">https://publicservices.international/affiliate/sutrel/videos?search=%7B%22types%22%3A%5B%5D,%22ally%22%3A%5B%5D,%22ally%22%3A%5B%5D%7D%7D&lang=en&affiliate\_id=708</a> (accessed 20 July 2022)
- Tamir, I., & Zoen, S. (2017) "Human Rights Impact Assessments in a Brazil Land Conflict: Towards a Hybrid Approach" *Business and Human Rights Journal*, Vol.2 No.2, pp. 371-377
- Taylor, M.B., Zandvliet, L., and Forouhar, M. (2009), "Due diligence for human rights: a risk-based approach", working paper no. 53, Corporate Social Responsibility Initiative, John F. Kennedy School of Government, Harvard University, Cambridge, MA, 1 September, available at: http://www.hks.harvard.edu/centers/mrcbg/programs/csri/research-publications/working-papers (accessed 20 July 2022).
- Teubner, G. (2006), "Substantive and Reflexive Elements in Modem Law" *Law & Society Review*, Vol. 17, No.2, doi: 10.2307/3053348
- Thame, D. (2019), "Recent Business and Human Rights Brazilian Regulation", *Beijing Law Review*, Vol.10, pp. 643-655, available at: <a href="https://doi.org/10.4236/blr.2019.104036">https://doi.org/10.4236/blr.2019.104036</a> (accessed 20 July 2022)
- The Landmark Project (2012, video file), "Verifying social standards in public procurement". A movie by "The LANDMARK Project", available at: https://www.youtube.com/watch?v=RFn1ceg9aGA (accessed 20 July 2022).
- Tommasi S., (2021), "The Liability of Internet Service Providers in the Proposed Digital Services Act', 29, European Review of Private Law, No. 6, pp. 925-944, https://kluwerlawonline.com/JournalArticle/European+Review+of+Private+Law/29.6/ERPL 2021048
- Thun Group of Banks (2013), "UN Guiding Principles on business and human rights. Discussion paper for banks on implications of Principles 16-21", available at: http://business-humanrights.org/sites/default/files/media/documents/thun-group-discussion-paper-final-2-oct-2013.pdf (accessed 20 July 2022).
- Umlas, E. (2009), "Corporate human rights reporting: an analysis of current trends", *Realizing Rights: The Ethical Globalization Initiative, UN Global Compact, Global Reporting Initiative*, available at: https://www.globalreporting.org/resourcelibrary/Human\_Rights\_analysis\_trends.pdf (accessed 20 July 2022).
- Unerman, J. and Zappettini, F. (2014), "Incorporating materiality considerations into analyses of absence from sustainability reporting", *Social and Environmental Accountability Journal*, Vol. 34 No. 3, pp. 172-186.

- United Kingdom of Great Britain and Northern Ireland, (2013), *Companies Act 2006* (Strategic Report and Directors' Report) Regulations 2013, available at: http://www.legislation.gov.uk/ukdsi/2013/9780111540169/contents (accessed 20 July 2022)
- United Kingdom of Great Britain and Northern Ireland, *Modern Slavery Act* (2015), available at http://www.legislation.gov.uk/ukpga/2015/30/contents/enacted (accessed 20 July 2022).
- United Nations Children's Fund & Global Child Forum, (2014) "Children rights and Business Atlas" <a href="https://www.childrensrightsatlas.org/">https://www.childrensrightsatlas.org/</a>, (last assessed 9 July 2022)
- United Nations Children's Fund (2014), "Children's rights in sustainability reporting. A guide for incorporating children's rights into GRI-based reporting", available at: http://www.unicef.org/csr/css/Childrens\_Rights\_in\_Reporting\_Second\_Edition\_HR.pdf (accessed 20 July 2022
- United Nations Environment Programme Finance Initiative (2014), "Human rights guidance tool for the financial sector", available at: http://www.unepfi.org/humanrightstoolkit/ ((accessed 20 July 2022).
- United Nations Environment Programme, Global Reporting Initiative, KMPG, and Centre for Corporate Governance in Africa (2013), "Carrots and sticks. Sustainability reporting policies worldwide today's best practice, tomorrow's trends", available at: https://www.globalreporting.org/resourcelibrary/Carrots-and-Sticks.pdf (accessed 20 July 2022)
- United Nations General Assembly (1948), *Universal Declaration on Human Rights* (10 December 1948, GA Res. 217A), available at: http://www.un.org/en/universal-declaration-human-rights/ (accessed 20 July 2022).
- United Nations General Assembly (1966a), *International Covenant on Civil and Political Rights*, available at: http://www.ohchr.org/en/professionalinterest/pages/ccpr.aspx (accessed 20 July 2022).
- United Nations General Assembly (1966b), *International Covenant on Economic, Social and Cultural Rights*, available at: http://www.ohchr.org/EN/ProfessionalInterest/Pages/CESCR.aspx (accessed 15 July 2022)
- United Nations General Assembly (2007), *Declaration on the rights of indigenous peoples* (2 October 2007, A/RES/61/295), available at: http://www.un.org/esa/socdev/unpfii/documents/DRIPS\_en.pdf (accessed 20 July 2022)
- United Nations General Assembly (2011) "Report of the Special Representative of the Secretary-General on the issue of human rights and transnational corporations and other business enterprises, John Ruggie. Addendum. Human rights and corporate law: trends and observations from a cross-national study conducted by the Special Representative", 23 May, available at: http://www.ohchr.org/Documents/Issues/Business/A-HRC-17-31-Add2.pdf (accessed 20 July 2022)
- United Nations General Assembly (2013), "Report of the Working Group on the issue of human rights and transnational corporations and other business enterprises. Addendum. Uptake of the Guiding Principles on Business and Human Rights: practices and results from pilot surveys of governments and corporations", available at:

  http://www.ohchr.org/Documents/HRBodies/HRCouncil/RegularSession/Session23/A-HRC-23-32-Add2\_en.pdf (accessed 20 July 2022)
- United Nations Global Compact (2010), "Guide to human rights impact assessment and management", available at: https://www.unglobalcompact.org/library/25 (accessed 20 July 2022)
- United Nations Global Compact (2016), *What is Global Compact*, available at https://www.unglobalcompact.org/what-is-gc/participants (accessed 20 July 2022)

- United Nations Global Compact (n.d.), "The Communication on Progress (COP) in Brief", available at https://www.unglobalcompact.org/participation/report/cop (accessed 15 July 2022)
- United Nations Global Compact and BSR (2014), "A guide to traceability. A practical approach to advance sustainability in global supply chains", United Nations Global Compact Office, New York, NY, available at: https://www.unglobalcompact.org/docs/issues\_doc/supply\_chain/Traceability/Guide\_to\_Traceability.pdf (accessed 10 July 2022)
- United Nations Global Compact (2022), "UN Global Compact Strategy 2021–2023", available at <a href="https://unglobalcompact.org/library/5869">https://unglobalcompact.org/library/5869</a>, (accessed 20 July 2022)
- United Nations Human Rights Council (2008): "Report of the Special Representative of the Secretary-General on the issue of human rights and transnational corporations and other business enterprises, John Ruggie", 7 April, available at: http://www.reports-and-materials.org/sites/default/files/reports-and-materials/Ruggie-report-7-Apr-2008.pdf (accessed 20 July 2022)
- United Nations Human Rights Council (2011), "Report of the Special Representative of the Secretary-General on the issue of human rights and transnational corporations and other business enterprises, John Ruggie. Guiding Principles on Business and Human Rights: Implementing the United Nations "Protect, Respect and Remedy Framework"", 21 March, available at: http://www.ohchr.org/Documents/Issues/Business/A-HRC-17-31\_AEV.pdf (accessed 20 July 2022)
- United Nations Human Rights Council (2014a), *Human rights and transnational corporations and other business enterprises* (23 June 2014, A/HRC/26/L.1), available at: http://ap.ohchr.org/documents/dpage\_e.aspx?si=A/HRC/26/L.1 (accessed 20 July 2022).
- United Nations Human Rights Council (2014b), "Elaboration of an international legally binding instrument on transnational corporations and other business enterprises with respect to human rights", (25 June 2014, A/HRC/26/L.22/Rev.1) available at: http://ap.ohchr.org/documents/dpage\_e.aspx?si=A/HRC/26/L.22/Rev.1 (accessed 20 July 2022)
- United Nations Human Rights Council, (2014c) 'Human Rights and Transnational Corporations and Other Business Enterprises', A/HRC/RES/ 26/22 (15 July 2014), http://daccess-dds-ny.un.org/doc/UNDOC/GEN/G14/083/82/PDF/G1408382.pdf?OpenElement (accessed 20 July 2022)
- United Nations Human Rights Council (2019), Gender dimensions of the Guiding Principles on Business and Human Rights: Report of the Working Group on the issue of human rights and transnational corporations and other business enterprises, available at: https://documents ddsny.un.org/doc/UNDOC/GEN/G19/146/08/PDF/G1914608.pdf?OpenElement(accessed 20 July 2022)
  United Nations Human Rights Council, (2021), "UNGPs 10+A Roadmap for the Next Decade of Business and Human Rights", available at: at
  - https://www.ohchr.org/sites/default/files/2021-12/ungps10plusroadmap.pdf (accessed 20 July 2022).
- United Nations Security Council (2010), "Resolution 1952 (2010). Adopted by the Security Council at its 6432<sup>nd</sup> meeting on 29 November 2010", 29 November, available at: http://www.un.org/ga/search/view\_doc.asp?symbol=S/RES/1952(2010) (accessed 20 July 2022)
- United Nations Sub-Commission on the Promotion and Protection of Human Rights

- (2003), "Norms on the Responsibilities of Transnational Corporations and Other Business Enterprises with Regard to Human Rights," (13 August 2003, E/CN.4/Sub.2/2003/12/Rev.2), available at:
- http://ap.ohchr.org/documents/alldocs.aspx?doc\_id=7440 (accessed 9 July 2022)
- United States Congress, (2022), "H.R.6256 To ensure that goods made with forced labor in the Xinjiang Uyghur Autonomous Region of the People's Republic of China do not enter the United States market, and for other purposes", available at https://www.congress.gov/bill/117th-congress/house-bill/6256/text (accessed 20 July 2022).
- United States Congress, 2010, *Dodd-Frank Wall Street Reform and Consumer Protection Act*(HR 4173, 5 January 2010), available at: https://www.sec.gov/about/laws/wallstreetreform-cpa.pdf (accessed 20 July 2022).
- United States Congress, (2022), *Uyghur Forced Labor Prevention Act, S.65* 117th Congress (2021-2022), available at https://www.congress.gov/bill/117th-congress/senate-bill/65 (accessed 20 July 2022).
- United States Treasury Department Office for Assets Control (2013), *Responsible Investment Reporting Requirements OMB NO*. 1405-0209, available at: http://www.humanrights.gov/wp-content/uploads/2013/05/responsible-investment-reporting-requirements-final.pdf (accessed 20 July 2022).
- United States Securities and Exchange Commission (2012), *Rule 13p-1*, available at http://www.sec.gov/rules/final/2012/34-67716.pdf (accessed 20 July 2022).
- Van Dijk, M., De Haas, M., & Zandvliet, R. (2018), "Banks and Human Trafficking: Rethinking Human Rights Due Diligence". *Business and Human Rights Journal*, Vol.3 No.1, pp.105-111.
- Veiberg C., Factor G. and Tedaldi R. (2019) "Measuring business impacts on human rights: "Practice and trends in the use of indicators for HRIA" in Gotzamann (Ed.) 2019, *Handbook on Human Rights Impact Assessment*. Edwards Elgar Publishing
- Wachenfeld, M., Wrzoncki, E. and de Angulo, L. (2019), Sector-wide impact assessment: A big picture approach to addressing human rights impacts in Gotzmann (Ed.), 2019, *Handbook on Human Rights Impact Assessment*, Edward Elgar Publishing, Cheltenham, pp. 85-186.
- Wagner, B. (2012), "Exporting censorship and surveillance technology", *Humanist Institute for Cooperation with Developing Countries*, available at: https://www.hivos.org/sites/default/files/exporting\_censorship\_and\_surveillance\_technology\_by\_ben\_wagner.pdf (accessed 20 July 2022)..
- Wagner, B. and Guarnieri, C. (2014), "German companies are selling unlicensed surveillance technologies to human rights violators and making millions", *Global Voices Advocacy*, 5 September, available at: https://advox.globalvoices.org/2014/09/05/exclusive-german-companies-are-selling-unlicensed-surveillance-technologies-to-human-rights-violators-and-making-millions/ (accessed 20 July 2022).
- Wakenge, C. Bashwira Nyenyezi, M., Bergh, S., Cuvelier, J., (2021), "From 'conflict minerals' to peace? Reviewing mining reforms, gender, and state performance in eastern Democratic Republic of Congo", *The Extractive Industries and Society*, Vol. 8, No. 2, https://doi.org/10.1016/j.exis.2021.100894.
- Wettstein, F.(2021), "Betting on the Wrong (Trojan) Horse: CSR and the Implementation of the UN Guiding Principles on Business and Human Rights", *Business and Human Rights Journal*, Vol. 6 No.2, p. 312-325, available at <a href="https://EconPapers.repec.org/RePEc:cup:buhurj:v:6:y:2021:i:2:p:312-325\_10">https://EconPapers.repec.org/RePEc:cup:buhurj:v:6:y:2021:i:2:p:312-325\_10</a> (accessed 20 July 2022).

- Wiss Y., Bensal T. (2019), "Knowing and showing: The role of HRIA in the food and beverage Sector", in Gotzmann (Ed.), 2019, *Handbook on Human Rights Impact Assessment*, Edward Elgar Publishing, Cheltenham, pp. 170-186.
- Wolfe, L. (2015), "How Dodd-Frank is Failing Congo", *Foreign Policy*, retrieved from: http://foreignpolicy.com/2015/02/02/how-dodd-frank-is-failing-congo-mining-conflict-minerals/ (accessed 20 July 2022).
- Wood, S. (2011), "The case for leverage-based corporate human rights responsibility", *Business Ethics Quarterly*, Vol. 22 No. 1, pp. 63-98.
- World Bank (2013), "Human Rights Impact Assessments: A Review of the Literature, Differences with other forms of Assessments and Relevance for Development", Commissioned by the Nordic Trust Fund, the World Bank available at <a href="https://documents1.worldbank.org/curated/en/834611524474505865/pdf/125557-WP-PUBLIC-HRIA-Web.pdf">https://documents1.worldbank.org/curated/en/834611524474505865/pdf/125557-WP-PUBLIC-HRIA-Web.pdf</a> (accessed 20 July 2022).
- World Benchmarking Alliance, (2022), "Social Transformation Baseline Assessment 2022", available athttps://assets.worldbenchmarkingalliance.org/app/uploads/2022/01/2022\_Social\_Transformation\_Baseline\_Assessment.pdf (accessed 20July 2022).
- Workers' Rights Consortium (2016), *The Workers' Rights Consortium*, available at: http://www.workersrights.org/ (accessed 20 July 2022).